Quality Education for Sustainable Development

An educator handbook for integrating values, knowledge, skills and quality features of Education for Sustainable Development in schooling

Includes a critical analysis of the Australian National Curriculum

by Dr. Joy M. de Leo





All Rights Reserved

© 2012 by J M de Leo

This work is copyright and may not be reproduced without the written consent of the author, in accordance with the Australian Copyright Act 1968.

First published in 2012 by UNESCO APNIEVE Australia Publishing, Adelaide South Australia Telephone: 61 (0)416 274177 Website: www.unesco-apnieve.edu.au Email: apnieve@adam.com.au

Paperback copies may be purchased from UNESCO APNIEVE Australia by ordering online at: www.qesd.org

National Library of Australia Cataloguing-in-Publication entry

Author: De Leo, J.M.

Title: Quality education for sustainable development [electronic resource]: an educator handbook for integrating values, knowledge, skills and quality features of education for sustainable development in schooling / Dr J M de Leo.

ISBN: 9780987403315 (ebook)

Notes: Includes bibliographical references and index.

Subjects: Education. Sustainable development.

Dewey Number: 370

A comprehensive set of Appendixes containing the research that informs and accompanies the content in this publication, may be downloaded at: URL: www.qesd.org/appendixes

Distributed by UNESCO APNIEVE Australia Text layout and Cover by Yooyah Pty Ltd

This book is based on a doctoral thesis presented to The University of Adelaide in South Australia for the degree of Doctor of Philosophy, for which a Dean's Commendation was awarded for Thesis Excellence.

The designations employed and the presentation of material throughout this publication do not imply the expression of any opinion whatsoever on the part of UNESCO or of UNESCO APNIEVE concerning the legal status of any country, territory, city or area or of its authorities, or the delimitation of its frontiers or boundaries.

The author is solely responsible for the choice and presentation of the contents of this book and for the opinions expressed therein, which are not necessarily those of UNESCO or of UNESCO APNIEVE and do not commit the organisations.

ABSTRACT

History shows that society looks to education to develop the values, knowledge and skills needed to address contemporary challenges and to create the kind of society and lifestyles that are appropriate to the time, place and context. This is particularly so in times of crisis and rapid change such as those encountered today. The inter-related issues of climate change, environmental degradation, globalisation and the global financial crisis (GFC), population explosion, growing intercultural and inter-religious conflict and increasing social concerns and inequities, confront education with many complex challenges.

This handbook provides a guide for educators on how these global challenges can be addressed by adopting a values-based approach to Education for Sustainable Development (ESD) in schooling. The research informing this handbook involved a thorough text and content analysis of key international standard-setting documents, reports and relevant scholarly research, to identify the values, knowledge, skills and key characteristics of a quality approach to ESD that have been consistently expressed. These are then synthesised to form a comprehensive and coherent framework of evaluative criteria for ESD, which are then used as the basis for analysing the first four learning areas of the Australian National Curriculum that had been released at the time of writing. This framework of evaluative criteria can be used for developing educational programs and curricula and reorienting educational systems towards ESD, as advocated by the *International Implementation Scheme for Education for Sustainable Development* (2004) – the key reference document informing ESD implementation.

This educator handbook on ESD is divided into three parts as follows:

Part one

Chapters 1, 2, 3 and 4 provide the introduction and background essential to the curriculum analysis. In particular, Chapter 1 outlines key global concerns driving the need for values-based Education for Sustainable Development (ESD) and points to potential approaches for addressing them through education. Chapter 2 discusses the nature of ESD as described in the *International Implementation Scheme for Education for Sustainable Development* (2004), which informs discussion throughout the handbook. Chapter 3 examines the nature of education across time, in particular the important role that values have played in education and clarifies definitions and understandings of values. Chapter 4 provides an overview of relatively recent initiatives in Australian schooling involving the integration of values, with particular relevance to ESD.

Part two

Chapters 5, 6 and 7 investigate the global values, knowledge, skills and quality characteristics needed in schooling to foster a peaceful, just and sustainable society. Specifically, Chapter 5 identifies and describes these global values and compares them to Australian values and to those in the *National Framework for Values Education in Australian schools* (NFVE), as well as to values that have endured across cultures for millennia. Chapter 6 reviews the educational thinking that led to current concepts of quality education and Education for Sustainable Development (ESD), with particular emphasis on values. It is argued that schools operate in three ways, by what they teach, by how they teach and by the kind of place the school is. These are examined in Chapter 6 in terms of quality and effectiveness for values-based teaching and learning for a sustainable future.

The key quality characteristics of education appropriate for ESD are discussed, involving values-based learning environments, relationships and other features. Chapter 7 identifies and discusses the knowledge and skills needed for ESD and highlights the need for trans-disciplinary approaches to curriculum development, in order to facilitate systems thinking and holistic problem solving, which are essential for sustainability. The quality characteristics of Education for Sustainable Development are summarised to form a set of criteria used to evaluate the Australian National Curriculum in Part three.

Part three

Chapters 8 to 12 present the outcomes of an in-depth analysis of the first four learning areas developed for the Australian National Curriculum for schooling, as a case study. Part three examines the presence in the curriculum of the values, knowledge, skills and quality characteristics for ESD that were identified in Part two, against the background context discussed in Part one.

The Australian National Curriculum is also tested against a philosophical and values-based framework for its coherence, consistency and strength and particularly, for its usefulness in creating a peaceful, just and sustainable society. The analysis examines the extent to which the Australian National Curriculum is designed to service the goals of sustainable development, given that the curriculum seeks to integrate sustainability as a cross-curriculum priority. Recommendations for curriculum enhancement are summarised in the concluding Chapter 13.

The extensive Appendixes that accompany this study can be accessed freely at this

url: www.qesd.org/appendixes

The Appendixes are designed to be referred to alongside the text, although some key tables are also inserted within the handbook for easy reference. In particular, readers are advised to refer to the following four key Appendixes that summarise the knowledge, skills, global values and characteristics needed for a

quality approach to ESD, which can be used as checklists to inform the development and evaluation of ESD programs and curricula:

Appendix 18 Global value sets appearing most frequently in international documents related to sustainable development. (Also listed in Table 5.1 in Chapter 5)

Appendix 32 Combined list of sustainability issues to inform ESD curriculum content.

Appendix 35 Combined list of sustainability skills to inform ESD curriculum development.

Appendix 40 Evaluative Criteria for values-based Education for Sustainable Development.

These Appendixes can be accessed at: www.qesd.org/appendixes

These Appendixes are referred to comprehensively in Part three, when analysing the extent to which the emerging Australian National Curriculum succeeds in integrating sustainability and meeting the objectives for ESD. Appendix 79 summarises the analysis of the Australian National Curriculum against the evaluative criteria for values-based ESD listed in Appendix 40.

It is argued that the issues addressed by Education for Sustainable Development (ESD) are too important to be under-represented in the Australian National Curriculum and merit serious and thorough consideration in the context of a rapidly changing World. Although this handbook specifically addresses school-based ESD, with a particular emphasis on values and less so on pedagogy and assessment, the ESD principles, values, knowledge, skills and quality characteristics contained in the four key Appendixes listed above, can be adapted for application to all levels of education.

TABLE OF CONTENTS

TITLE PAGE

i

DEDICATION ACKNOWLEDGEMENTS		
		PREFACE
ABSTRACT	xii	
TABLE OF CONTENTS	xv	
TABLES AND FIGURE		
PART ONE		
CHAPTER 1: LIVING AND LEARNING IN A RUNAWAY WORLD		
Introduction	1	
The power of ideas	4	
The contribution of the United Nations		
Towards a National School Curriculum		
Conclusion	14	
CHAPTER 2: EDUCATION FOR SUSTAINABLE DEVELOPMENT		
Introduction	15	
What is Education for Sustainable Development?		
The evolving concept of sustainable development		
The UN Decade of Education for Sustainable Development (DESD)		
Clarifying terms used to describe ESD	20	
The evolution of Education for Sustainable Development	21	
Implementing the UN Decade of Education for Sustainable Development	24	
Values-based Education for Sustainable Development	24	
Multiple interconnected dimensions of Education for Sustainable Development	26	
Links between the UN Decade for ESD and other UN initiatives	29	
Convergence of values-based educational areas towards ESD		
Conclusion		

Introduction		
The nature and purpose of education		
Understanding the nature of values		
Clarifying values-related terms	39	
Types and categories of values	42	
Values in education or education in values?	48	
Values education across time and cultures	50	
Why values education, now?	56	
The response of education to global challenges	61	
Conclusion	70	
CHAPTER 4: VALUES IN AUSTRALIAN SCHOOLING		
Introduction	71	
Values in education in Australia	71	
Values in the National Goals for Schooling		
Values in Australian State and Territory educational jurisdictions	73	
Education in values in Australia	74	
The National Framework for Values Education in Australian Schools	75	
Quality criteria for good practice values education in Australian schooling	76	
Evolution of Education for Sustainability (EfS) in Australia	78	
Australian Sustainable Schools Initiative (AuSSI)	79	
AuSSI and whole school approaches to Education for Sustainability (EfS)		
Global Education in Australian Schooling		
Conclusion	82	
PART TWO		
CHAPTER 5: GLOBAL VALUES FOR A SUSTAINABLE WORLD		
Introduction	83	
Global values	83	
Meaning of global values		
Analysis of global values		
Origins of global values		
Comparison between global and Australian values		

Reconciling diverse values		
Conclusion		
CHAPTER 6: QUALITY TEACHING AND LEARNING FOR SUSTAINABLE DEVE	ELOPMENT	
Introduction	107	
Quality Education for Sustainable Development		
Supportive learning environments, relationships and modelling values	109	
Values-based learning for Education for Sustainable Development	112	
Developing the whole learner	117	
Education for societal change and transformation	123	
Engaging the whole school community in Education for Sustainable Development	126	
Conclusion	128	
CHAPTER 7: CURRICULUM FOR A SUSTAINABLE WORLD		
Introduction	129	
Knowledge for Education for Sustainable Development	129	
Skills for Education for Sustainable Development	132	
Integrated, trans-disciplinary curriculum for Education for Sustainable Developmen	t 137	
Evaluative criteria for values-based Education for Sustainable Development	144	
Conclusion	148	
PART THREE		
TAKI IIIKEE		
CHAPTER 8: INTRODUCING THE AUSTRALIAN NATIONAL CURRICULUM		
Introduction	150	
Research methods	150	
Background	153	
Educational goals in the Australian National Curriculum	154	
Structure and elements of the Australian National Curriculum	156	
Learning areas	156	
Years of schooling	156	
General capabilities	157	
Cross-curriculum priorities	159	
Diverse learners		
Assessment against achievement standards		

Definition of Quality in the Australian National Curriculum	
Conclusion	165
CHAPTER 9: SKILLS IN THE AUSTRALIAN NATIONAL CURRICULUM	
Introduction	167
Skills in the Australian National Curriculum	167
Skills in the Australian National Curriculum relevant to sustainability	172
Summary of skills in the Australian National Curriculum relevant to sustainability	180
Conclusion	181
CHAPTER 10: KNOWLEDGE IN THE AUSTRALIAN NATIONAL CURRICULUM	
Introduction	182
Knowledge in the Australian National Curriculum relevant to sustainability	182
Global Perspectives in the Australian National Curriculum	183
Environmental Issues in the Australian National Curriculum	184
Environmental Sustainability in the National Science curriculum	185
Environmental Sustainability in the National History curriculum	187
Environmental Sustainability in the National English curriculum	188
Environmental Sustainability in the National Mathematics curriculum	189
Summary of relevant knowledge in the Australian National Curriculum	191
Conclusion	194
CHAPTER 11: VALUES IN THE AUSTRALIAN NATIONAL CURRICULUM	
Introduction	196
Values in the Australian National Curriculum	198
References to specific values in the Australian National Curriculum	200
Global values in the curriculum	203
Values relating specifically to sustainability	204
Conclusion	206
CHAPTER 12: EVALUATING THE AUSTRALIAN NATIONAL CURRICULUM	
AGAINST THE CRITERIA FOR ESD	
Introduction	209
Multidimensionality in the Australian National Curriculum	209
Curriculum integration and the Australian National Curriculum	
Developing the whole learner in the Australian National Curriculum	

whole school community approach in the Australian National Curriculum	
Evaluative criteria for ESD in the Australian National Curriculum	
Conclusion	217
CHAPTER 13: CONCLUSION	
Summary of findings regarding the Australian National Curriculum	219
Potential for improvements to the Australian National Curriculum	220
Implications for teacher education	222
Implications for further research	
REFERENCES	228
ACRONYMS	256
LIST OF APPENDIXES	259

TABLES AND FIGURE

Table 2.1	Values expressed in the International Implementation Scheme (IIS) for the UN Decade of Education for Sustainable Development	26
Table 2.2	The cultural perspectives of ESD as described in the IIS (2004)	28
Table 2.3	The scope of the four dimensions of Education for Sustainable Development	28
Table 2.4	The characteristics of Education for Sustainable Development (ESD) as described in the International Implementation Scheme (IIS)	33
Chapter 3		
Table 3.1	Description of Haidt and Joseph's Moral Classification System	46
Table 3.2	Characteristics of the UNESCO four pillars of learning	65
Table 3.3	The Seven Tensions of the Delors report	66
Chapter 4		
Table 4.1	Values in the National Framework for Values Education (NFVE) in Australian schools	75
Table 4.2	Scope of Learning in the Australian Sustainable Schools Initiative (AuSSI)	80
Chapter 5		
Table 5.1	Global value sets found in international documents	85
Table 5.2	Global values mapped against Maslow's Hierarchy of Needs	96
Table 5.3	A synthesis of Australian values from various sources	103
Chapter 6		
Table 6.1	Most common features of quality education synthesised from relevant international and Australian documents	111
Table 6.2	Whole person competencies for ESD	119
Chapter 7		
Table 7.1	ESD skills in the International Implementation Scheme (IIS)	133
Table 7.2	Skills for ESD identified by McKeown	134
Table 7.3	A paradigm shift for education in moving towards ESD in the Asia Pacific region	145
Table 7.4	Seven Principles in Australia's 2009 Action Plan for EfS	146

	PART ONE	
Figure 5.1	World Value Survey Cultural Map (1999-2004)	93
CHAPTER 5		
	FIGURE	
Table 12.1	Frequency of references relating to each ESD dimension in the Australian National Curriculum	210
Chapter 12		
Table 11.3	Summary of references to the terms 'values', 'morals', 'ethics' 'attitudes', 'principles' and 'beliefs' in the Australian National Curriculum	206
Table 11.2	Instances of 'interdependence' and 'interconnectedness' in the Australian National Curriculum	205
Table 11.1	Values in the Australian National Curriculum	197
Chapter 11		
Table 10.1	Frequency of references relating to each ESD dimension in the Curriculum	185
Chapter 10		
Table 9.3	Interconnected understandings in the Australian National Curriculum	177
Table 9.2	References to the future in the curriculum relevant to sustainability	175
Table 9.1	Most frequent skill groupings in the Australian National Curriculum	168
Chapter 9		
Table 8.3	References in the Australian National Curriculum to the sustainability of practices used by other cultures	161
Table 8.2	References in the Australian National Curriculum to the environmental Practices of ATSI peoples	161
Table 8.1	Intercultural understanding in the Australian National Curriculum	160
Chapter 8		
Table 7.6	The 14 categories of evaluative criteria for ESD	148
Table 7.5	quality education approaches appropriate for ESD	147

LIVING AND LEARNING IN A RUNAWAY WORLD1

Introduction

Never in recorded history has the fate of humanity depended on the actions of human beings as it does today. Over the past few decades, the World has undergone unprecedented rapid change in almost every area of human activity and interaction, bringing new opportunities and emerging challenges. Indeed, more change has occurred by human hand in the last century than in millennia. For millions of years, the natural, evolutionary processes of nature that carved out the geographic features of the planet and led to the immense diversity of plant and animal life, including humans and their socio-cultural structures, seemingly have been overtaken by the scientific, technological and informational capacities of the human mind. While there is cause for celebration in human achievement, the speed and scale of change demand cautious deliberation, as humanity struggles to come to terms with the unanticipated and perilous consequences of progress and the rapidly expanding population of the World.

In times of change society has always looked to education to develop the knowledge, skills and values needed to address contemporary challenges and to create the kind of society and lifestyle appropriate to the time, place and context. This has been particularly so in times of crisis, transition and rapid change such as those encountered today. Dewey and others adapted Plato's view that it was first necessary to know the desired 'end' purpose of existence and the type of society needed, for education to be designed accordingly:

Unless we know the end, the good, we shall have no criterion for rationally deciding what the possibilities are which should be promoted. (Dewey, 1916, p. 88)

A strong international consensus has emerged around "the end" that is universally desired for "the good" of all and what is required of education to respond to current circumstances in all areas of life, socio-cultural, socio-political, socio-economic and environmental, consistent with globally shared values. The term most commonly used to describe this is 'Education for Sustainable Development' (ESD) that is discussed in Chapter 2.

This study examines the interdependent nature of contemporary challenges and the ways in which values-based Education for Sustainable Development (ESD) can respond in practice for the development of a peaceful, just and sustainable World at a time of great global change.

¹ Title adapted from Giddens, A. 2000. Runaway World. London: Routledge.

There is considerable evidence of both the benefits and threats of progress in all fields of human endeavour. For example, advances in medicine and technology have brought relief and well-being to the quality of life of millions, having eased suffering, cured and alleviated disease, reduced infant mortality, increased life expectancy and provided fresh water, electricity and education to many parts of the World. Nevertheless they have also led to a population explosion. Recent developments in science and biogenetics continue to improve health and well-being, but further exacerbate population growth and give rise to new threats, inequalities and attendant ethical dilemmas.

The advances in communication technologies and transport have accelerated social and economic globalisation and interdependence, spread democratic ideals, extended human rights and multiplied opportunities for intercultural and knowledge exchange. Such advances have brought people from the remotest parts of the World into closer proximity with others than ever before. Yet some aspects of progress have also brought unexpected effects, such as a growing digital divide, the rapid loss of linguistic and cultural diversity furthering cultural standardisation and the rising social, racial and religious conflicts that threaten human security.

Economic globalisation has provided wealth to many, but poverty to many more, creating both winners and losers leading to enormous wealth on the one hand and extreme poverty and inequality on the other, both within and between countries. The spread of democracy has brought freedom and opportunities for civic participation never seen before on such a large scale. However, this period has also been accompanied by the growth of a 'runaway' global market economy dominated by large trans-national institutions, many of which exceed the wealth and power of most nations.

Despite global efforts, the persistent problems of over-population, poverty, illiteracy, hunger, disease and inequity have remained of concern throughout the World and in many places have worsened (UNDP, 2007, p. 2). Layered over these are new challenges, such as increasing violence and terrorism, growing numbers of refugees seeking asylum, severe environmental degradation, climate change, biodiversity loss, increasing natural disasters, famine and the threat of pandemic from which no-one is immune. Giddens called this "manufactured risk", with which there is little experience, since this is the first time in history that humanity has become concerned about what it has done to nature (Giddens, 2000).

The panoply of global problems currently experienced spans all dimensions of human life, socio-cultural, political, economic and environmental, interwoven in a complex web of interdependencies in ways that are impossible for experts and leaders alone to solve. International cooperation and interdisciplinary exchange are crucial if humanity is to succeed in solving the challenges that threaten the very continuation of life on Earth. Such cooperation requires education to prepare future leaders and researchers in cross-disciplinary

engagement. It is increasingly difficult to shut out problems by strengthening border controls - a modern version of the walled fortifications of the past. The reach of globalisation, electronic communication, climate change and other global concerns transcend national boundaries, making protectionist efforts futile and cooperation essential. In this context it is imperative that education foster the requisite global values to achieve such cooperation and exchange.

With such tremendous intellectual and technological firepower, humankind has the capacity either to create universal well-being or to continue damaging what nature has taken millions of years to create. Humanity might also continue to provoke conflict over ideologies or resources, while some enjoy excess and others suffer deprivation, without the global values to temper such excesses. Instead, future scenarios might be shaped by what is valued and considered most important, whether it be: (a) individual accumulation of wealth, or collective human well-being; (b) proving one ideology over another, or respecting diverse world views; and (c) the glory of winning and prevailing over others, or harmonious unity with nature and with others while protecting ecological and cultural diversity. The Nobel laureate economist Amartya Sen argued that shared norms and social values underpinned the kind of life that everyone would want and that freedom and democratic processes provided the context within which shared values could be negotiated through dialogue (Sen, 1999a). Such shared values, when agreed globally, as they are in international agreements and conventions, can form the basis for living together equitably, harmoniously and sustainably on Earth, using education to reinforce them.

Yet it seems that the qualities of the human heart, such as values, compassion and connection with nature, have been unable to keep pace with the human mind and its capacity for technological and scientific progress. These are evident in all areas of life including education. Western emphasis on reason and logic that began with Plato and Aristotle, was championed by Descartes in the seventeenth century and continued by Kant in the eighteenth century, even if spawning such immense scientific progress, might have led circuitously to a deficit of values today. As a result of increasing urbanisation and the commercial production of food on a large scale, humanity has also experienced a disconnection from the natural world upon which it has always depended for identity and survival. Society appears to have forgotten that humans are not separate from nature, instead they are an intimate part of natural ecosystems. Such systems are characterised by a delicately balanced interdependence of flora, fauna and resources, sustained by natural water and carbon cycles that have taken millennia to evolve and just a few centuries to dismantle.

While the Western emphasis on reason has led to immense technological and scientific progress, it has not been balanced by the application of values, integral to the feeling, sensing and intuitive side of human nature. The Nobel laureate and neural scientist Sperry pinpointed neglect of human values as the "primary

underlying cause of most of our difficulties" (Sperry, 1983, p. 9). Saul also saw the "void in society" as being caused by an "absence of values" (Saul, 1993, p. 584). Recent developments in neural science and moral psychology have revealed the positive influences of perception, intuition and emotion on choice, values and priorities (Haidt and Joseph, 2004). Failing to nurture these in education can foster the development of individuals who base their decisions on reason alone, at a cost to society and to the environment. The neural scientist, Damasio (1994) demonstrated that the use of reason alone, unsupported by emotion, might cause even greater problems than faulty logic or irrationality. In the field of education, Hill claimed that, "technical values associated with economic rationalism were inadequate to keep the peace and maintain social structures", while arguing for a values education focus on social justice and shared community goals (Reynolds, 2001, p. 23). Einstein also lamented the erosion of ethical values, caused by an overemphasis on fact and intellect, with serious consequences for humanity (Einstein, 1951).

The power of ideas

The pervasive influence of Cartesian philosophy offers just one example of the profound impact that the books of great thinkers, philosophers, scientists and educators have had on human endeavour, some positive, others not. Armstrong's substantial analysis of texts dating back to the Axial Age almost three thousand years ago, revealed considerable similarity in emerging moral codes across Eastern, Western, Indian and Middle Eastern civilisations, suggesting that at least some human values, such as compassion for example, might be integral to human nature (Armstrong, 2006). Some neural scientists and psychologists were also of this view (Dawkins 1976, Pugh 1977, Sperry 1983, Pinker 2003). A rich source of human values that shaped thinking for millennia is contained within the ancient Indian *Vedas*, the Confucian *Analects*, Aristotelian *Ethics*, Buddha's *Kanjur*, the Babylonian *Talmud*, the Christian *Bible* and the Islamic *Qur'an*. Armstrong also found that these texts emerged, for the most part, at times of great social change (Armstrong, 2006), not unlike the profound socio-cultural, but also climatic-environmental changes experienced today on a much larger and broader scale.

It has only been during the past few centuries that prevailing ideas have shifted, at least in the West, leading to dramatic changes in the ways people live, work and relate to nature. We are therefore at a turning point in history, ready for the emergence of new values and the renewal of old ones, transmitted through education and communication technologies. In this book, an investigation of the thinking of educators and philosophers across time and cultures reveals that the development and transmission of values have always been part of the aims of education, particularly in times of great change. With the current accelerating rate of change in an interdependent World, the many issues demanding attention and cooperation locally and globally, require education to foster values-based thinking and inspirational ideas in learners that can motivate them to take positive, competent and informed action.

A study of past societies revealed that values were reinforced strongly in times of rapid change and turmoil. Given the widespread changes of significant magnitude that have occurred over the past 50 years and the lifestyle transformations likely in the decades to come necessitated by climate change, the strengthening of values is sure to be needed once again, shaped by current circumstances. In order to facilitate this period of transition, Armstrong suggested that the ideas of the past should be revisited, to "adapt the original insights ... to the needs and circumstances of today" (Armstrong, 2006, p. xvii). Sperry also recommended the adoption of values suited to the needs of the times:

The obvious recommendation is to shape up our value systems to something more in tune with present-day reality, more properly suited to the new powers that man now commands and the new problems we now face. (Sperry, 1983, p. 10)

It helps to know where humanity has been, but also to have a compass for guiding preferred directions for the future. As with the texts of ancient times in the past, the great books of modern times influence thinking across all dimensions of human life and fields of study today. They help in the search for potential solutions to present and future realities, but also raise key questions that stimulate thinking and beg inquiry.

In the **natural** dimension of life, Darwin's treatise *On the Origin of Species* (Darwin, 1859), while controversial both then and now (since it challenged religious tenets of creationist origins of the Earth) nonetheless provided the foundations for evolutionary biology through the theory of natural selection and adaptation and revealed the inherited origins of diversity through natural processes. This diversity is now threatened by human hand. In recent times there have been attempts made to remove the study of evolution from the subject of Biology in American and even in Australian schools, by creationists who have attributed evolution to intelligent design rather than to natural processes. This trend is disturbing since it gives credence to sectarian belief and ideology, not to be confused with values, rather than to Science. As history has shown, ideological clashes cause bitter conflict, but Science has the potential to free human-kind from the yoke of ideology, balanced by ethics and shared values for a better World for all. An overemphasis on reason does not warrant a reactive descent into ideology and irrationality. Agreed and shared values towards collective well-being have the potential to create a pivotal point of balance between reason and illogic, reinforced by an education system that harmonises cognitive with physical and socio-emotional development and which clearly distinguishes between values, beliefs and knowledge.

Neural scientists, biologists, psychologists and moral philosophers alike share an interest in evolutionary biology in seeking to resolve the contentious nature-nurture debate and the extent to which moral sense is inherent, imprinted genetically, or socialised. Recent discoveries relating to neural plasticity have clearly

indicated that both nature and socialisation play their part (Doidge, 2007), with implications for values-based childrearing and education.

In the realm of science, Watson's discovery of the structure of DNA, described in his autobiographical account The Double Helix (Watson, 1968), led to a proliferation of genetic research and discoveries culminating in the mapping of the entire human genome in 2000. The discovery was hailed an historic landmark achievement and "the most wondrous map ever produced by humankind" (BBC, 2000). This powerful knowledge of the key to human creation previously accorded to an omnipotent God alone, while bringing enormous benefit compels scientists to question the ethical considerations and possible speciespolluting consequences of unchecked human interference in natural evolutionary processes. The risks and potential consequences of the genetic manipulation of plant, animal and human genomes, if allowed to escape laboratory confines, can exceed the capacity of the most learned scientists to control. Runaway genetic change can alter the path of evolution forever, the consequences of which are entirely unknown. The United Nations (UN) system establishes international standard-setting documents for addressing such risks, such as the Universal Declaration on the Human Genome and Human Rights (UNESCO, 1999a) and the Universal Declaration on Bioethics and Human Rights (UNESCO, 2006), replete with inspiring value-laden phrases and informed by existing human rights conventions. Nowhere is the need for a rigorous application of values and ethics more evident than in the field of genetic science and research, which has already found its way into school and university curricula, necessitating the integration of values in its educational approach.

The **economic** dimension of society is also in need of values to balance the rampant economic rationalism, over-production and consumerism, that have dominated the wasteful 'throw-away' culture for so long with dire consequences for so many. Aristotle referred to the **economy** as the "wise government of the family" (Horne, 1992, p. 23), but today economists have strayed far from this view of the family of humanity, obsessed with continuous growth, profit and productivity at all costs in a market motivated by greed (Horne, 1992, pp. 8-9). Saul referred to the "ethical slide" of the market and to a "marginalization of ethics" paralleled by an intense rise in corruption (Saul, 2006, pp. 184-185). Hamilton maintained that economics relied solely on the "validity of rationality to the exclusion of all else" (Hamilton, 1994, p. 160), leading to a lack of consideration for human well-being and environmental integrity. Although a strong and stable economy can bring material well-being and access to health, education, an enjoyable life and other benefits, economic production is fed from natural environmental resources, without which economic activity would falter. Alternative and sustainable economic systems are needed that can deliver universal well-being, while preserving the resources and natural systems upon which humanity depends. This cannot occur while the existing economic paradigm remains unchallenged and unchecked by global values, despite its repeated failure to deliver equitable benefits to the majority.

Free market ideologies that had promised to deliver prosperity, have instead led to widespread economic depression twice in 80 years, causing poverty, hardship and loss for many and an unstable global market economy that threatens to fail once again. Hawken claimed that while ecologists were concerned with nature and economists were concerned with economics, "human beings are abandoned to the marketplace" (Hawken, 1993, p. 131), unmitigated by a global values framework that would maintain a sense of justice, fairness and order. History has shown that economic disparity, exclusion and injustice inevitably lead to social conflict. Hence, it would be wise to observe the call of Voltaire to make the "dominant passion ... the public weal" (Saul, 1993, p. 6). Hawken called for an economy directed to creating "the best life for the greatest number of people" (Hawken, 1993, p. 205). In the same vein, Sperry called for a commitment to progress in terms of "improvement in the quality and dimensions of life" for all (Sperry, 1983, p. 23).

In his seminal work, *Development as Freedom* (Sen, 1999a), Sen interconnected economic with social and political freedoms as the path to universal well-being and highlighted the important role of education and democracy in achieving this. Sen explored the cross-dimensional aspects of development and advocated the expansion of individual and collective freedoms, (political, social, protective and economic), as both the primary end and principal means of development. Sen maintained that these alleviated human deprivation and oppression and provided choice and opportunity, thereby maximising human capacity to act as agents of positive change (Sen, 1999a, pp. 36-38). He argued that the traditional focus on economic and technological development and markets as the only **means** for human advancement had been unsuccessful in delivering equitable or humane outcomes, whereas the expansion of freedoms as an overarching objective "directs attention to the ends (i.e. human well-being) that make development important" (Sen, 1999a, p. 3). This view of development as needing to be human-centred rather than economically-centred is also advocated by the UN, which regularly calls for the humanising of development and for "globalization with a human face" (UNDP, 1999).

In the **socio-political** and **educational** dimensions of life, Connell considered Dewey's 1916 treatise, *Democracy and Education*, the most important educational text of the twentieth century (Connell, 1980, p. 72). Dewey conducted a critical analysis of contemporary theories of knowing and moral development, which were to influence educators for over 80 years. He proposed an "intimate organic connection" between the acquisition of knowledge, everyday life experiences, both inside and outside school, and educational methods and materials with character development, moral growth and conduct (Dewey, 1916, p. 360). Dewey connected learning with participation in everyday life, involving realisations about values through experiencing rather than merely by theoretical learning, socialisation, or routine habit formation.

Dewey proposed that education in a democracy was ideally to be orderly and sequenced, forming a developmental social process that enabled free communication, equal and informed participation, contributing towards societal change and well-being and leading to the continued capacity for learning and growth throughout life (Dewey, 1916). Education was not to be limited to the classroom, nor crammed into 12 brief years, but to extend to the whole school community and throughout life. Education was to be immediately relevant to everyday experiences, events and emotionally charged interactions, thereby ensuring their retention in memory. From the work of Dewey, Sen and numerous relevant UN documents, it is concluded that education in a free and open democratic society that promotes shared values for societal well-being, can assist greatly in addressing socio-economic injustice, resolving conflict and restoring balance with the natural world.

While democracy has adopted various forms and levels throughout the World, it is generally associated with individual and collective civic participation, rights, liberties and freedom of speech, in varying degrees. According to Giddens (2000), democracy was "the most powerful energising idea of the twentieth century", having more than doubled in the number of countries that have adopted democracy throughout the World since the mid-1970s. Some countries have had to struggle and fight for democracy, while others have experienced transition problems or reversals, while in oppressive regimes human rights have frequently been abused (Giddens, 2000). However, Saul recalled that long-standing democracies in the West themselves also "emerged slowly out of highly imperfect conditions" (Saul, 2006, p. 203). Democratic values and principles are therefore to be nurtured and reinforced constantly through education to maintain their original intent in practice, or at least towards forms of governance that retain respect for human rights and well-being.

Australia inherited its democratic system from Britain without having to fight for it, thus many freedoms are now taken for granted. More is expected of democracy and a sense of disillusionment with democratic process is growing, accompanied by a loss of trust in politicians who are perceived as self-interested and corrupt. Giddens explained this trend as occurring in open information societies that produced "more active, reflexive citizenries" in a "detraditionalising world". To counter it would require a deepening and broadening of democracy at the global level (Giddens, 2000) and the promotion of democratic principles and values in education (such as 'participation' and 'open dialogue') reflected in the content, methods and processes.

Although considerable attention has been given to authoritative texts in the sciences, socio-politics, economics and education, this has not been the case in the field of **ecology** until recently. Early warnings were expressed in the environmental classics, such as: Marsh's *Man and Nature* (1864); Carson's *Under the Sea-Wind* (1941), *The Sea Around Us* (1951) and *Silent Spring* (1962), which helped trigger the beginnings of

the environmental movement; Commoner's *The Closing Circle* (1971); and *The Limits to Growth* report to The Club of Rome (Meadows, Randers and Behrens, 1972), which predicted that economic growth could not continue indefinitely because of the limited availability of natural resources. Yet only committed environmentalists responded to these until the United Nations (UN) organisation initiated a series of influential international conferences and reports that captured public attention.

Gore's book and documentary film of the same title, *An Inconvenient Truth* (Gore, 2006), highlighted the shocking impact of human activity on climate, which for the first time reached a global audience of millions. The predictions of social and economic collapse, foreseen in *The Limits to Growth* report (1972) appeared to be unfolding. The Intergovernmental Panel on Climate Change (IPCC, 2007), in its fourth assessment report, *Climate Change 2007*, gave evidence of melting glaciers, rising seas, changing weather and temperature patterns, inundation, transforming landscapes, extreme events, reduced rainfall, drought and water and food insecurity, among other findings, as a result of human-induced warming. Because of the level of controversy surrounding climate change, the Royal Society published *Climate Change: A Summary of the Science* (Pethica et al., 2010) and the Australian Academy of Science published *The Science of Climate Change: Questions and Answers* (AAS, 2010), which challenged the sceptics who sought to delay action on climate change for at least two decades. Although some nations are gradually developing national policies and strategies to address climate change, the level of international cooperation leaves much to be done, since political, economic and corporate interests continue to dominate concerns.

The contribution of the United Nations

Arguably the system best able to foster the level of international cooperation needed to address the significant challenges faced by humanity is the United Nations (UN), with almost universal representation among nations. The founders of the UN left a structural legacy after World War II that today enables open international debate and agreements in many fields based upon principles drawn from the UN founding documents, such as the 1945 *UN Charter* and the 1948 *Universal Declaration of Human Rights* (UDHR). Since its very beginnings the UN has fostered global dialogue, exchange and cooperation towards common goals based on shared values, directed to the equitable advancement of humanity in all areas of life. Collective aspirations for a peaceful, just and sustainable World have been expressed in the numerous international conventions and agreements to which UN member countries are signatories. This book analyses the global values most commonly expressed in relevant documents as a basis for application to Education for Sustainable Development (ESD) in concert with local values.

For decades the UN has convened international conferences that have addressed a wide range of socioeconomic and environmental concerns, beginning with the 1972 Stockholm Conference on the Human Environment, followed later by the 1987 Brundtland World Commission on Environment and Development (WCED) and more recently, the UN Climate Change conferences held in Copenhagen in 2009, in Cancun in 2010 and in Durban in 2011, to name a few. The UN also declared the period 2005 to 2014 the UN Decade of Education for Sustainable Development (DESD), with the goal of integrating the principles, values and practices of sustainable development into all aspects of education and learning. The DESD aimed to foster changes in behaviour towards a more sustainable future in the areas of the environment, the economy and society, in ways appropriate to local cultural contexts. This decade coincided with the International Decade for a Culture of Peace and Non-violence for the Children of the World (2001-2010), the International Year of Planet Earth Triennium (2007-2009) and the UN Decade for Human Rights Education (1995-2004), which led to the establishment of the World Programme for Human Rights Education (WPHRE). These international initiatives together provide a concentrated international focus on the type of education needed for a peaceful, just and sustainable World and an opportunity for change not to be missed. It remains to be seen whether the global rhetoric translates into practical action.

By analysing the content of these and other relevant international documents, this study demonstrates that what is needed at this time is an integrated, values-based approach to Education for Sustainable Development (ESD) that develops the whole learner (i.e. cognitive, physical, emotional, social, moral and spiritual). This handbook focuses particularly on the development of values for ESD since values are very much needed at this time and have been somewhat neglected in education when compared to the acquisition of knowledge and skills. Nonetheless, the key features of a quality approach to ESD are also examined, including the knowledge and cognitive and functional skills needed for transforming thinking, lifestyles and work practices, tested against the Australian National Curriculum as a case study.

Even though pedagogy is addressed only in passing in this book, the need for conducive learning environments and positive relationships are emphasised, as are democratic processes of teaching and learning to foster responsible civic participation and informed ethical action for bringing about positive societal change. Education has a special role to play in facilitating the transformation needed, which is to shift from an individualist and materialist consumer society to one that embraces both individual and collective needs for a sustainable future, while preserving the environment and enabling well-being for all.

The subject is addressed from a global perspective applied to the Australian context however the principles and processes advanced can be adapted to any national or regional context. No country can isolate itself from global concerns in an interdependent and globalised World and learners today need to be prepared to become global citizens while addressing local concerns. Although this book applies the values, knowledge, skills and characteristics for ESD to schooling, these are equally adaptable to all levels of education and training.

The key agency in the UN system with responsibility for education and for international exchange in education, the sciences, culture and communications, is the United Nations Educational, Scientific and Cultural Organization (UNESCO). UNESCO is also the leading agency for implementing the UN Decade of Education for Sustainable Development (2005-2014), which promotes an integrated approach to education for sustainability based on values, bringing together all interconnected dimensions of sustainability (i.e. socio-cultural, environmental and economic). This handbook draws on the 2004 draft International Implementation Scheme (IIS) for the UN Decade of Education for Sustainable Development (ESD) as the guiding document to inform the development of key characteristics of ESD, needed to address global challenges for a sustainable World.

Since its establishment, UNESCO has published numerous influential books and documents relevant to this subject, promoting values-based solutions to socio-cultural, economic and environmental issues, for attaining equity, peace, human rights and well-being for all. Among these was de Cuéllar's 1995 Report of the World Commission on Culture and Development (WCCD), *Our Creative Diversity*, which reformulated the concept of development with a **socio-cultural** focus, in terms of human well-being rather than of economic progress alone. The report proposed alternative models for development that preserved cultural heritage and diversity based on a global ethic of shared values. The report was significant since it directed sustainability towards maintaining economic viability, protecting environmental integrity and diversity, sustaining social cohesion and preserving cultural and linguistic diversity; an aspect of sustainability that often has not been recognised.

The field in which UNESCO has been most influential is undoubtedly **Education**, both internationally and in Australia. Two of the most influential international reports on education to emerge from UNESCO were: (a) Faure's 1972 Report of the International Commission on the Development of Education, *Learning to Be: The world of education today and tomorrow*, which proposed whole-person lifelong learning for "democracy, humanistic development and change" for redressing the imbalanced "relationship between man and his environment" (Faure, 1972, p. 101); and (b) the Delors 1996 Report of the International Commission on Education for the Twenty-first Century, *Learning: The Treasure Within*, which also highlighted the need for whole learner development through values-based education (Delors, 1996, p. 94). This book discusses the findings of these two leading education reports and other relevant UNESCO publications insofar as they relate to ESD.

In writing about UNESCO's purpose and philosophy after its founding in 1946, UNESCO's then Director-General Mayor cited Huxley, who argued that values must consciously guide the work of the organisation: UNESCO cannot be neutral in the face of competing values ... any ... system (of values) which is unconsciously assumed is less likely to be true than one which is consciously sought after and studied. (Mayor, 1995, p. 79)

It is argued that the conscious and explicit choice of global values are needed in education for a sustainable World, without which: (a) market-based and media-imposed values might continue to dominate without regard for human well-being or environmental concerns; (b) scientists and bio-geneticists might continue their developmental research untempered by ethical or environmental considerations; and (c) the homogenising impact of communication technologies might reduce the world's rich and diverse tapestry of cultures and languages to a handful that hold economic and technological sway.

A founding and active member of UNESCO, Australia is a young nation that has not experienced war within its borders, is located on an island continent in the southernmost part of the globe, and is seemingly sheltered from global vicissitude. Nonetheless, the reality of global interdependence has now reached its shores. The inter-related issues of climate change, environmental degradation, the global financial crisis (GFC), growing intercultural and inter-religious conflict in a multicultural society, increasing pressure from asylum seekers, the constant threat of terrorism and growing social concerns and inequities internally, present education in Australia with many complex challenges and competing priorities. A consistent national approach to education for a sustainable society across the Australian States and Territories is required to address these challenges. Such an imperative applies to all nations for addressing respective local challenges linked to global concerns.

Towards a National School Curriculum

There has been a tradition of innovation in education in Australia dating back to the mid-1960s when the first ideas for a National School Curriculum were being explored by educational pioneers ahead of their time. Although the six separate colonies had formed a national Federation of States in 1901, the Australian States and Territories retained legislative control over public education and the school curriculum, as they do today, making consensus difficult (Reid, 2005, p. 15). The only effective national forums for professional exchange in the 1960s and early 1970s were the education conferences organised under the auspices of the Australian National Advisory Committee for UNESCO, since official curriculum cooperation was difficult. It was not until 1973 when the Commonwealth Schools Commission was established, followed by the Curriculum Development Centre (CDC) in 1974, that national cooperation in education officially began (Keeves, 1999, pp. 120-121). There were repeated attempts at establishing a national curriculum by several Federal Education Ministers, beginning with Fraser in 1968 with the Australian Science Education Project (ASEP), Dawkins between 1988 and 1993 and by Nelson in 2003. Efforts were also made by Skilbeck from the CDC in 1980 and by the Curriculum Corporation with the production of national learning materials from

1997. But it was not until mid-2010 that the first national curriculum documents were officially released for implementation. After a journey of more than 40 years, the Australian National School Curriculum had finally arrived. It had been helped along the way by the first statement of the National Goals for Schooling in 1989, which was revised in 1999 and again in 2008 and by the development of the Core Curriculum for Australian Schools by the Curriculum Development Centre (CDC) in 1980 and the mapping of the Mathematics Curriculum across all States and Territories in 1988, followed by Science and Technology (Keeves, 1999, p. 122).

Until very recently values and moral reasoning were not explicitly evident in Australian school curricula or in the various learning domains, except for a sub-area of personal development (Keeves, 1999, p. 122), since education was for a long time considered to be either values-free or values-neutral (Keown, 2005, p. 1). The Values Education Study commissioned by the Australian Government in 2003 found that schools were not values-free and that formal and hidden curricula were in fact based on values (DEST, 2003). The Australian Government then established the National Framework for Values Education in Australian Schools (NFVE), conducting trials in selected Australian schools and documenting case studies of good practice, leading to quality criteria for values education. This work provided the foundation for the systematic integration of values in Australian schooling.

Although the development and integration of values in education can trigger the necessary personal and societal changes of attitude and provide the intrinsic motivation to adopt sustainable lifestyles, values alone are insufficient to create the extent of transformation required for a sustainable society. A paradigm shift is needed that includes values, but that also alters systems of thinking by critically examining underlying assumptions, challenging old habits and developing a systemic global consciousness that understands the interconnections between the local and the global. Trans-disciplinary and transformative learning are vital for enabling critical systems thinkers to find innovative, ethical solutions to complex, inter-related and multi-dimensional problems from diverse perspectives and with equitable outcomes.

Given the urgency of current problems that threaten our socio-cultural, environmental and economic systems, learners and educators need not only to undergo a personal transformation of values and systems of thinking, but also need to understand the holistic processes of personal and societal transformation that engage the whole person. By understanding personal change processes, learners acquire the tools to develop themselves continually throughout life. Furthermore, by understanding social change processes they are empowered to become agents of change in their school, community and work place. In addition to acquiring values and transforming thinking, learners therefore also need to develop the functional skills to apply knowledge and values in practice for collective benefit.

These are just some of the factors that need to be considered when developing educational programs and curricula for sustainability. Although the Australian National Curriculum is used as a case study in this handbook, the principles, features and processes outlined here may be applied to ESD programs and curricula in any country adapted to local concerns.

Conclusion

From an extended text analysis of international documents and relevant literature, this book examines the values, knowledge, skills and quality features of Education for Sustainable Development (ESD) and considers how values can best be taught and learned for a sustainable future. Downey (1965) claimed that schools operate in three ways: by what they teach, by how they teach and by the kind of place the school is. These three areas are investigated to propose a coherent approach to values-based ESD, built upon a comprehensive review of scholarly writing and international documents that set the standards required. In addition to the values, skills and knowledge for ESD, the quality features of ESD are advanced to form of a set of evaluative criteria, tested against the developing Australian National Curriculum as a case study - the first Australian curriculum deliberately to include 'sustainability' as a cross-curriculum priority (Commonwealth of Australia, 2010, p. 5).

This handbook emphasises the need for education at all levels to enable learners to become responsible and ethical global citizens, equipped to contribute to the societal transformation needed for sustainable human development. Such an education would ideally be based on the shared global values derived from the UN system that can foster a peaceful, just and sustainable World. Since international consensus has already been reached regarding the nature of education needed to respond to current circumstances, termed Education for Sustainable Development (ESD), this study therefore begins with the background to ESD and the International Scheme (IIS) proposed for its global implementation, hereafter referred to respectively as ESD and the IIS.

REFERENCES

(AAS) Australian Academy of Science. 2010. *The Science of Climate Change: Questions and Answers.* Canberra: Australian Academy of Science. [Online, accessed 25 February 2011].

URL: http://www.science.org.au/policy/climatechange.html

(ABS) Australian Bureau of Statistics. 2006. *Cultural Diversity Overview*. Canberra: Commonwealth of Australia. [Online, accessed 27 August 2011]. URL:

http://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/C724250359785DC6CA25754C0013DC0A/\$File/20700_cultural_overview.pdf

(ACARA) Australian Curriculum, Assessment and Reporting Authority. 2010a. *The Shape of the Australian Curriculum (Version 2.0)*. Sydney: ACARA. [Online, accessed 19 January 2011].

URL: http://www.acara.edu.au/verve/_resources/Shape_of_the_Australian_Curriculum.pdf

(ACARA) Australian Curriculum, Assessment and Reporting Authority. 2010b. *The Australian Curriculum Science (Version 1.1)*. Sydney: ACARA. [Online, accessed 19 January 2011].

URL: http://www.acara.edu.au/curriculum/curriculum.html

(ACARA) Australian Curriculum, Assessment and Reporting Authority. 2010c. *The Australian Curriculum History (Version 1.1)*. Sydney: ACARA. [Online, accessed 19 January 2011].

URL: http://www.acara.edu.au/curriculum/curriculum.html

(ACARA) Australian Curriculum, Assessment and Reporting Authority. 2010d. *The Australian Curriculum Mathematics (Version 1.1).* Sydney: ACARA. [Online, accessed 19 January 2011].

URL: http://www.acara.edu.au/curriculum/curriculum.html

(ACARA) Australian Curriculum, Assessment and Reporting Authority. 2010e. *The Australian Curriculum English (Version 1.1)*. Sydney: ACARA. [Online, accessed 19 January 2011].

URL: http://www.acara.edu.au/curriculum/curriculum.html

(ACARA) Australian Curriculum, Assessment and Reporting Authority. 2010f. *Report on Trial School Consultation*. Sydney: ACARA. [Online, accessed 19 January 2011].

URL: http://www.acara.edu.au/curriculum/curriculum.html

(ACARA) Australian Curriculum, Assessment and Reporting Authority. 2010g. *Curriculum Development Process (Version 4.0)*. Sydney: ACARA. [Online, accessed 19 January 2011].

URL: http://www.acara.edu.au/curriculum/curriculum.html

(ACER) Australian Council for Educational Research. 2001. *Attitudes and Values Questionnaire*. Melbourne: ACER. [Online, accessed 12 July 2011]. URL: http://www.acer.edu.au/documents/AVQ_Brochure.pdf

Adams, D. 1993. "Defining Educational Quality." Arlington, VA: Institute for International Research. (*IEQ Publication No. 1. Biennial Report.*) ADEA 2003. [Online, accessed 19 April 2010]. URL: http://www.unicef.org/girlseducation/files/QualityEducation.PDF

(AEC) Australian Education Council. 1989. *The Hobart Declaration on Schooling: Common and Agreed National Goals for Schooling in Australia*. Hobart: Australian Education Council. [Online, accessed 15 July 2010].

URL: http://www.mceecdya.edu.au/mceecdya/hobart_declaration,11577.html

(AI) Amnesty International. 2007. *Amnesty International Report 2007. The state of the world's human rights.* London: Amnesty International. [Online, accessed 25 September 2009].

URL: http://archive.amnesty.org/report2007/

Aikin, W. M. 1942. *Adventure in American Education Volume 1. The Story of the Eight-year Study.* New York: Harper & Brothers.

Ainley, J. et al. 1998. Schools and the Social Development of Young Australians. Melbourne: ACER Press.

Anderson, J. 1997. "Content and text analysis." *Educational research, methodology, and measurement: an international handbook.* J.P. Keeves, ed. Oxford: Pergamon, 340-344.

APNIEVE. 1998. Learning to Live Together in Peace and Harmony. A Sourcebook for Teachers. Bangkok: UNESCO Publishing.

APNIEVE. 2002. Learning to Be. A Holistic and Integrated Approach to Values Education for Human Development. Bangkok: UNESCO Publishing.

APNIEVE. 2005. Learning to Do. Values for Learning and Working Together in a Globalised World. Bonn, Germany: UNEVOC.

Armstrong, K. 2006. The Great Transformation. London: Atlantic Books.

Arnold, R. 2005. Empathic Intelligence. Sydney: University of New South Wales.

Austin, A. G. 1972. Australian Education 1788-1900. Carlton, Victoria: Pitman.

Australian Education Council. (AEC). 1989. *The Hobart Declaration on Schooling: Common and Agreed National Goals for Schooling in Australia*. Hobart: Australian Education Council.

[Online, accessed 15 July 2010]. URL:

http://www.mceecdya.edu.au/mceecdya/hobart_declaration,11577.html

Australian Citizenship Council. 2000. *Australian Citizenship in a New Century*. Canberra: Commonwealth of Australia.

Banathy, B. H. ed. 1995. "Developing a Systems View of Education." *Educational Technology, June 1995.*USA.

Bateson, G. 1979. Mind and Nature. A Necessary Unity. New York: Bantam Books.

British Broadcasting Commission (BBC). 2000. "Scientists crack human code." *BBC News. 26 June 2000*. London: BBC. [Online, accessed 2 July 2010]. URL: http://news.bbc.co.uk/2/hi/science/nature/805803.stm

Beare, H. and Slaughter, R. 1993. Education for the Twenty-First Century. London: Routledge.

Bliss, S. 2007. *Global Education in Secondary Commerce, Economics, Business Studies, Legal Studies, History and Geography Syllabuses.* New South Wales: University Consortium. [Online, accessed 15 August 2009]. URL:

http://www.globaleducation.edna.edu.au/globaled/go/engineName/filemanager/pid/123/UWS_HSIE_GlobalEducation.doc?actionreq=actionFileDownload&fid=21280

Bloom, B. S. 1956. Taxonomy of Educational Objectives. Boston, MA.: Allyn and Bacon.

Bloom, B. S., Hastings, J. T. and Madaus, G. F. 1971. *Handbook on Formative and Summative Evaluation of Student learning*. New York: McGraw Hill.

Boudon, R. 2001. The Origin of Values. Sociology and Philosophy of Beliefs. USA: Transaction Publishers.

Boutros-Ghali, B. 1993. *Opening address to the 1993 Vienna World Conference on Human Rights*. Vienna: United Nations. [Online, accessed 20 January 2008].

URL: http://www.unhchr.ch/html/menu5/d/statemnt/secgen.htm

Brundtland, G. H. 1987. *The Report of the World Commission on Environment and Development (WCED):*Our Common Future. Oxford UK: Oxford University Press.

Bryk, A. S. and Schneider, B. 2002. *Trust in schools: A core resource for improvement*. New York: Russell Sage Foundation.

Calder, M. and Smith, R. 1991. A Better World for All: Development Education for the Classroom. Canberra: AGPS, AIDAB.

Carson, R. L. 1941. Under the Sea-Wind. New York: Oxford University Press.

Carson, R. L. 1951. The Sea Around Us. New York: Oxford University Press.

Carson, R. L. 1962. Silent Spring. New York: Houghton Mifflin Company.

Cavanagh, R. F. 2002. *Educational Visioning Processes: A Modern Footnote on an Ancient Task?* 2002 Annual Conference of the Australian Association for Research in Education. Brisbane: AARE. [Online, accessed 25 July 2010]. URL: http://www.aare.edu.au/02pap/rey02432.htm

Centre for Ecoliteracy (CEL). 2008. *Sustainability and Schools*. USA: CEL. [Online, accessed 5 June 2008]. URL: http://www.ecoliteracy.org/education/schools.html

Colins, C. and Chippendale, P. 1995. New Wisdom II. Values-Based Development. Brisbane: Acorn Publications.

Commoner, B. 1971. The Closing Circle: Nature, Man, and Technology. New York: Knopf.

Commonwealth of Australia. 1994. *Civics and Citizenship Education. Report of the Civics Expert Group.* Canberra: AGPS.

Commonwealth of Australia. 1995. *The Australian Constitution*. Canberra: Australian Government Publishing Service (AGPS).

Commonwealth of Australia. 1999. *A New Agenda for Multicultural Australia*. Canberra: Australian Government Publishing Service (AGPS). [Online, accessed May 2010].

URL: http://www.immi.gov.au/media/publications/multicultural/agenda/agenda1.htm

Commonwealth of Australia. 2000. *Environmental Education for a Sustainable Future: National Action Plan.* Canberra: Environment Australia. [Online, accessed 27 July 2010].

URL: http://www.environment.gov.au/education/publications/nap/pubs/nap.pdf

Commonwealth of Australia. 2002. *Global Perspectives. A Statement on Global Education for Australian Schools.* Melbourne, Australia: Curriculum Corporation.

Commonwealth of Australia. 2005. *Values for Australian Schooling Professional Learning Resources, Primary*. Canberra: Curriculum Corporation.

Commonwealth of Australia. 2006. *Implementing the National Framework for Values Education in Australian Schools. Report of the Values Education Good Practice Schools Project – Stage 1. Final Report.* Victoria: Curriculum Corporation. [Online, accessed March 2009].

URL: http://www.curriculum.edu.au/verve/_resources/VEGPS1_FINAL_REPORT_081106.pdf

Commonwealth of Australia. 2008. At the Heart of What We Do: Values Education at the Centre of Schooling – The Final Report of the Values Education Good Practice Schools Project – Stage 2. Victoria: Curriculum Corporation. [Online, accessed March 2009].

URL: http://www.curriculum.edu.au/verve/_resources/VEGPSP-2_final_3.pdf

Commonwealth of Australia. 2009a. *Living Sustainably. The Australian Government's National Action Plan for Education for Sustainability.* Canberra: Department of the Environment, Water, Heritage and the Arts. [Online, accessed 1 May 2010]. URL: http://www.environment.gov.au/education/publications/pubs/national-action-plan.pdf

Commonwealth of Australia. 2010. *Sustainability Curriculum Framework. A guide for curriculum developers and policy makers.* Canberra: Department of Education, Employment and Workplace Relations (DEEWR). [Online, accessed 25 April 2011]. URL: http://www.environment.gov.au/education/publications/pubs/curriculum-framework.pdf

Connell, W. F. 1980. A History of Education in the Twentieth Century World. Canberra: Curriculum Development Centre (CDC).

Costanza, R. ed. 1991. *Ecological Economics: The Science and Management of Sustainability.* New York: Columbia University Press.

Council of the Parliament of the World's Religions. (CPWR). 1993. *Declaration Toward a Global Ethic.* Chicago: CPWR. [Online, accessed 13 March 2008].

URL: http://www.cpwr.org/_includes/FCKcontent/File/TowardsAGlobalEthic.pdf

Council of the Parliament of the World's Religions (CPWR). 1999. *A Call to Our Guiding Institutions*. Cape Town: CPWR. [Online, accessed 13 March 2008].

URL: http://www.cpwr.org/_includes/FCKcontent/File/CalltoGuidingInstitutions.pdf

CSIRO. 2011. Climate Change: Science and Solutions for Australia. Melbourne: CSIRO Publishing.

Curriculum Corporation. 1998. Voices and Values: Citizenship in Asia. Melbourne: Curriculum Corporation.

Curriculum Corporation. 2008. *Global Perspectives. A Framework for Global Education in Australian Schools.*Melbourne, Australia: Curriculum Corporation.

Curriculum Council of Western Australia (CCWA). 2005. Values in the Curriculum. Relationship between Core Values identified in the Western Australian Curriculum Framework and the Australian Government's Values for Australian Schooling. Perth: CCWA. [Online, accessed 25 July 2010].

URL: http://www.curriculum.wa.edu.au/internet/Years_K10/Values_in_the_Curriculum_Framework

Damasio, A. R. 1994. "Descartes' Error and the Future of Human Life." *Scientific American, October 1994*. [Online, accessed 2 July 2010].

URL: http://www.docstoc.com/docs/2591794/Descartes-Error-and-the-Future-of-Human-Life

Darwin, C. R. 1859. *On the origin of species by means of natural selection, or the preservation of favoured races in the struggle for life.* London: John Murray. 5th edition. [Online, accessed October 2010].

URL: http://darwin-online.org.uk/content/frameset? itemID=F387 & viewtype=text & pageseq=1.

Darwin, C. R. 1872. The Expression of the Emotions in Man and Animals. New York: Appleton.

Dawkins, R. 1976. The Selfish Gene. Oxford: Oxford University Press.

DECS. 2005. *Statement of Directions (2005-2010)*. Adelaide, South Australia: DECS. [Online, accessed 25 July 2010]. URL: http://www.decs.sa.gov.au/aboutdept/default.asp?id=23712&navgrp=163

DECS. 2006. Does Spiritual Wellbeing Belong in Education? Spiritual Well-being and Education Discussion Paper. Adelaide, South Australia: Department of Education and Children's Services (DECS).

DECS. 2007. A Guide to becoming a Sustainable School. South Australia: Department of Education and Children's Services (DECS). [Online, accessed 17 August 2009]. URL: http://www.decs.sa.gov.au/efs/

de Cuéllar, P. 1995. *Our Creative Diversity. Report of the World Commission on Culture and Development.*Paris: UNESCO Publishing. [Online, accessed 8 October 2007].

URL: http://www.UNESCO.org/culture/policies/ocd/html_eng/action8.shtml

DEET. 2005. Valuing the EsseNTials. The Northern Territory Project implementing the National Values Education Framework. Northern Territory: Department of Employment, Education and Training (DEET).

DEH. 2005. Educating for a Sustainable Future. The National Environmental Education Statement for Australian Schools. Melbourne: Curriculum Corporation. [Online, accessed 14 March 2008]. URL: http://www.environment.gov.au/education/publications/pubs/sustainable-future.pdf
DEH. 2006. Comparative Assessment: Australian Sustainable Schools Initiative Pilot Programme in New South Wales and Victoria. Canberra: Department of the Environment and Heritage (DEH).

DEH. 2007. Caring for Our Future. The Australian Government Strategy for the United Nations Decade of Education for Sustainable Development, 2005–2014. Canberra: Department of the Environment and Heritage (DEH). [Online, accessed May 2011]. URL: http://www.environment.gov.au/education/publications/pubs/caring.pdf

DEH. 2008. AuSSI Sustainable Schools website. [Online, accessed 14 June 2008].

URL: http://www.environment.gov.au/education/aussi/resources.html

de Leo, J. M. 2005. "Values in Education". Proceedings of the International Symposium, *Cultivating Wisdom, Harvesting Peace*. 10-13 August 2005, Brisbane: Griffith University.

Delors, J. 1996. Learning: The Treasure Within. Report of the International Commission on Education for the Twenty-first Century. Paris: UNESCO Publishing.

DEST. 2003. *Values Education Study. Final Report*. Melbourne: Curriculum Corporation. [Online, accessed 26 July 2010]. URL: http://www.valueseducation.edu.au/verve/_resources/VES_Final_Report14Nov.pdf

DEST. 2005. *National Framework for Values Education in Australian Schools*. Canberra: Commonwealth of Australia. [Online, accessed July 2010].

URL: http://www.curriculum.edu.au/verve/_resources/Framework_PDF_version_for_the_web.pdf

DEST. 2005a. *DEST Strategic Plan (2005-2008)*. Canberra: Department of Education, Science and Technology. [Online, accessed 3 July 2011]. URL: http://www.dest.gov.au/NR/rdonlyres/ADE17372-B489-485F-9B74AB5BFB9377FD/8490/StrategicPlanconcept_FINAL_WEB.pdf

DET ACT. 2006. *Strategic Plan (2006-2009)*. Canberra: Australian Capital Territory Department of Education and Training. [Online, accessed 3 July 2011].

URL: http://www.det.act.gov.au/__data/assets/pdf_file/0015/17502/stratplan2006_2009.pdf

DET NSW. 2004. *Values in NSW Public Schools*. Sydney: New South Wales Department of Education and Training. [Online, accessed 25 July 2010].

URL: https://www.det.nsw.edu.au/media/downloads/languagesupport/school_values/english.pdf

DET NT. 2009. Strategic Plan (2009-2012): Delivering a Smart Territory through Quality Education and Training. Darwin: NT Department of Education and Training. [Online, accessed 26 July 2010].

 $URL: http://www.det.nt.gov.au/__data/assets/pdf_file/0013/4126/DET_strategic_plan2009-2012.pdf$

DET WA. 2008. Plan for Public Schools (2008-2011). Perth, WA: Department of Education and Training.

DETA. 2008. *Strategic Plan 2008-2012*. Brisbane: Queensland Department of Education Training and the Arts. [Online, accessed 25 July 2010].

URL: http://education.qld.gov.au/strategic/planning/pdfs/strategic-plan-08-12.pdf

Dewey, J. 1916. Democracy and Education. Canada: Macmillan.

DEWHA. 2010. *Australian Sustainable Schools Initiative. Case Studies.* Canberra: Department of the Environment, Water, Heritage and the Arts (DEWHA). [Online, accessed 28 July 2010]. URL: http://www.environment.gov.au/education/aussi/about.html

DIC. 2007. Statement of Australian Values. Canberra: Department of Immigration and Citizenship (DIC). [Online, accessed 15 March 2008]. URL: http://www.immi.gov.au/living-in-australia/values/statement/long/index.htm

DoE. 2000. *Tasmanian Curriculum: Values, Purposes and Goals.* Tasmania: Department of Education. [Online, accessed 25 July 2010]. URL: http://www.education.tas.gov.au/curriculum/values

Doidge, N. 2007. *The Brain that changes itself.* New York: Silberman books.

Downey, L. W. 1965. The Secondary Phase of Education. New York: Blaisdell Publishing.

ECI. 2000. *The Earth Charter.* The Netherlands: Earth Charter Initiative (ECI). [Online, accessed 8 October 2007]. URL: http://www.earthcharterinaction.org/2000/10/the_earth_charter.html

Einstein, A. 1951. "The Need for Ethical Culture." *Speech celebrating the seventy-fifth anniversary of the Ethical Culture Society.* 5 January 1951, New York: Ethical Culture Society.

Einstein, A. 1954. *Ideas and Opinions*. US: Crown publishers.

Ellyard, P. 2001. Ideas for the New Millennium. Victoria: Melbourne University Press.

Evers, C. W. 1999. "From Foundations to Coherence in Educational Research." *Issues in Educational Research*. pp. 264-278. Oxford: Pergamon.

Evers, C. W. and Lakomski, G. 1991. *Knowing Educational Administration*. Oxford: Pergamon.

Faure, E. 1972. Report of the International Commission on the Development of Education, Learning to Be: The world of education today and tomorrow. Paris: UNESCO.

Ferreira, J., Ryan, L., Davis, J., Cavanagh, M. and Thomas, J. 2009. *Mainstreaming Sustainability into Preservice Teacher Education in Australia*. Canberra: Australian Research Institute in Education for Sustainability (ARIES) for the Department of the Environment, Water, Heritage and the Arts.

Flowers, C. 2002. Values and Civic Behaviour in Australia. The National Engagement Project Trial. Melbourne: Brotherhood of St Laurence.

Fullan, M. 2005. Leadership and Sustainability. California: Corwin Press.

Furco, A. 2006. *Unpacking the Nature of Values Education in Primary School Settings*. Berkely: International Center for Research on Civic Engagement and Service-Learning, University of California. [Online, accessed 6 April 2008]. URL: http://education.qld.gov.au/curriculum/values/docs/andy-furco.ppt

Galbally, F. 1978. Review of Post-Arrival Programs and Services to Migrants. Canberra: AGPS.

Gardner, H. 1999. Intelligence Reframed: Multiple Intelligences for the 21st Century. New York: Basic Books.

Gardner, H. 2011. Truth, beauty, and goodness reframed: Educating for the virtues in the 21st century. New York: Basic Books.

Georgescu, D. 1997. *Global Education Charter. Working document.* Bucharest: Institute for Educational Sciences.

Giddens, A. 2000. Runaway World. London: Routledge.

Gilligan, C. 1982. *In a different voice: Psychological Theory and Women's Development.* Cambridge MA: Harvard University Press.

Glasersfeld, E. 1996. *Homage to Jean Piaget (1896-1980)*. Massachusetts: University of Massachusetts. [Online, accessed 23 August 2008]. URL: http://www.oikos.org/Piagethom.htm

Glendon, M. 2001. A World Made New: Eleanor Roosevelt and the Universal Declaration of Human Rights. New York: Random House.

Goleman, D. 1995. Emotional Intelligence. Great Britain: Bloomsbury.

Goleman, D. 2006. Social Intelligence. The New Science of Human Relationships. New York: Bantam.

Goodman, L. E. 2003. *Islamic Humanism*. London: Oxford University Press.

Gore, A. 2006. An Inconvenient Truth: The planetary emergency of global warming and what we can do about it. New York: Rodale.

Gore, J. 1998. *Revisiting Values Education: The Role of Civics and Citizenship Education.* Mexico: Pacific Circle Consortium Conference Papers.

Habermas, J. 1990. Moral Consciousness and Communicative Action. Cambridge, Massachusetts: MIT Press.

Haidt, J. and Joseph, C. 2004. "Intuitive Ethics: How Innately Prepared Intuitions Generate Culturally Variable Virtues." *Daedalus. Fall 2004*, pp. 55-66, Cambridge, MA: MIT Press.

Haidt, J. 2007. "Moral psychology and the misunderstanding of religion" *Edge, The Third Culture 2007*. USA: Edge Foundation. [Online, accessed 28 August 2011].

URL: http://www.edge.org/3rd_culture/haidt07/haidt07_index.html

Haidt, J. 2008. "Morality." *Perspectives on Psychological Science*. 3, pp. 65-72. Washington, DC: Association for Psychological Science.

Hall, B. P. 1994. *Values Shift. A Guide to Personal and Organisational Transformation*. Rockport: Twin Lights Publishers.

Halstead, J. M. and Taylor, M. J. 1996. Values in Education and Education in Values. London: Falmer Press.

Halstead, J. M. and Taylor M. J. 2000. "Learning and Teaching about Values: A Review of Recent Research." *Cambridge Journal of Education*. Vol. 30, No. 2. Cambridge UK: Cambridge University Press.

Hamilton, C. 1994. The Mystic Economist. Canberra: Willow Park Press.

Harbom, L. ed. 2006. *States in Armed Conflict*. Uppsala, Sweden: Uppsala University, Department of Peace and Conflict Research.

Harris-Hart, C. 2009. *The National History Curriculum: Tragedy or Triumph?* Biennial Australian Curriculum Studies Conference, Canberra: School of Education, Deakin University. [Online, accessed 22 February 2011]. URL:http://www.historyteacher.org.au/htdocs/national_curriculum/ACSA%202009%20Paper%20Catherine %20Harris%20Hart.pdf

Hawken, P. 1993. The Ecology of Commerce. New York: HarperCollins.

Hawkes, N. 2003. How to inspire and develop positive values in your classroom. Cambridge: LDA.

Hay, D. and Nye, R. 2006. The Spirit of the Child. London: Kingsley Publishers.

Henderson, K. and Tilbury, D. 2004. Whole school Approaches to Sustainability: An International Review of Sustainable School Programs. Sydney: Australian Research Institute for Education for Sustainability (ARIES).

HENT. 2005. *Trans-disciplinary Inquiry incorporating holistic principles*. Hobart: Holistic Education Network of Tasmania (HENT). [Online, accessed 22 April 2008]. URL: http://www.hent.org/transdisciplinary.htm

Hicks, D. and Holden C. ed. 2000. *Teaching the Global Dimension. Key Principles and Effective Practice*. London and New York: Routledge Taylor and Francis Group.

Hill, B. V. 2004. Exploring Religion in School: A National Priority. Adelaide: Open Book Publishers.

Hoepper, B. 2002. *Peace Education and Studies of Society and Environment*. Occasional paper prepared for the Queensland School Curriculum Council. Brisbane: Faculty of Education QUT.

Holdren, J. P., Daily, G. C., and Ehrlich, P. R. 1995. *The Meaning of Sustainability: Biogeophysical Aspects.*Washington DC: United Nations University-World Bank.

[Online, accessed 26 May 2011]. URL: http://dieoff.org/page113.htm

Horne, D. 1992. The Trouble with Economic Rationalism. Melbourne: Scribe, Monash University.

Horne, D. 1997. The Avenue of the Fair Go. Australia: HarperCollins Publishers.

Hughes, P. and More, A. J. 1997. "Aboriginal Ways of Learning and Learning Styles." Paper presented at the Annual Conference of the Australian Association for Research in Education, 1997. Brisbane: AARE. [Online, accessed 10 June 2010]. URL: http://www.aare.edu.au/97pap/hughp518.htm

Hume, D. 1739. *A Treatise of Human Nature*. Indiana, USA: Quantonics. [Online, accessed 9 September 2010]. URL: http://www.quantonics.com/Hume_AToHN_B1_P4_S2_of_SRS.html

IBE. 2001. "Education for All for Learning to Live Together." 46th Session of the International Conference of Ministers of Education (ICE) Conference Papers. Geneva: International Bureau of Education (IBE). [Online, accessed 8 October 2007]. URL: http://www.ibe.UNESCO.org/International/ICE/46english/46finrape.htm

Inglehart, R. and Welzel, C. 2005. *Modernization, Cultural Change and Democracy*. New York: Cambridge University Press. [Online, accessed 8 August 2012].

URL: http://www.worldvaluessurvey.org/wvs/articles/folder_published/article_base_54

IPCC. 2007. *Climate Change 2007: Synthesis Report*. Contribution of Working Groups I, II and III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC). Geneva, Switzerland: IPCC. [Online, accessed 15 July 2010]. URL: http://www.ipcc.ch

IUCN. 1980. *The World Conservation Strategy: Living Resource Conservation for Sustainable Development.*Gland, Switzerland: International Union for the Conservation of Nature (IUCN), UNEP and WWF.

IUCN. 1991. Caring for the Earth: A Strategy for Sustainable Living. Gland, Switzerland: IUCN, World Conservation Union, UNEP, World Wildlife Fund for Nature.

Kahneman, D. 2011. Thinking, fast and slow. Australia: Penguin Group.

Kant, I. [1785], Denis, L. ed. 2005. Groundwork of the Metaphysics of Morals. Canada: Broadview Editions.

Keeves, J. P. 1999 "Research into Curriculum Change." *Australian Education: Review of Research 1965-1998*. Melbourne: ACER.

Keown, P. et al. (2005) *Values in the New Zealand Curriculum: A Literature Review on Values in the Curriculum.* Wellington: New Zealand Ministry of Education. [Online, accessed 23 October 2007]. URL: http://www.minedu.govt.nz/index.cfm?layout=document&documentid=7010&indexid=8854&indexparentid=1004

Kessler, R. 2000. *The Soul of Education*. Virginia, USA: Association for Supervision and Curriculum Development (ASCD).

Kiernan, J. 2005. *Values Education in Australian Schools: Support materials for Tasmanian Schools.* Tasmania: Department of Education. [Online, accessed 25 July 2010].

URL: http://www.curriculum.edu.au/verve/_resources/Values_Education_support_Tas.pdf

King-Boyes, M. 1977. Patterns of Aboriginal Culture: Then and Now. Sydney, Australia: McGrawHill.

Kirschenbaum, H. 1992. "A Comprehensive Model for Values Education and Moral Education." *Phi Delta Kappan, v73 n10 p77176, June 1992.* USA: Phi Delta Kappan.

[Online, accessed 19 March 2008]. URL: http://www.hi-ho.ne.jp/taku77/refer/kirsch.htm

Kohlberg, L. 1958. *The Development of Modes of Thinking and Choices in Years 10 to 16.* Ph.D. dissertation, University of Chicago, Illinois.

Kohlberg, L. and Higgins, A. 1987. "School democracy and social interaction." Kurtines, W. M. and Gewirtz, J. L. eds. *Moral development through social interaction*. New York: John Wiley and Sons.

Kolb, D. 1984. *Experiential Learning: Experience as a Source of Learning and Development*. Englewood Cliffs, NJ: Prentice Hall.

Lemin, M., Potts, H. and Welsford, P. eds. 1994. *Values Strategies for Classroom Teachers*. Victoria: Australian Council for Educational Research (ACER).

Leu, E. and Price-Rom, A. 2005. *Quality of Education and Teacher Learning: A Review of the Literature.* USA: American Institutes for Research and Academy for Educational Development.

Lewy, A. 1968. "The empirical validity of major properties of a taxonomy of affective educational objectives." *The Journal of Experimental Education, Vol. 36, No. 3 (Spring, 1968), pp. 70-77.* [Online, accessed 14 December 2011]. URL: http://www.jstor.org/pss/20156973

Lincoln, S. F. 2006. *Challenged Earth. An Overview of Humanity's Stewardship of Earth.* London: Imperial College Press.

Lovat, T. and Toomey, R. eds. 2007. *Values Education and Quality Teaching. The Double Helix Effect.* New South Wales: David Barlow Publishing.

Lovell, P. 2006. "Assessing values education – is it possible?" *Independent Education, Vol. 36 No. 3, November 2006*. [Online, accessed 12 July 2011].

URL: http://www.curriculum.edu.au/verve/_resources/Lovell_Independent_Education.pdf

Mackay, H. 2007. Advance Australia... Where? Sydney: Hachette Australia.

Marcus, G. F. 2004. The Birth of The Mind: How a Tiny Number of Genes Creates the Complexities of Human Thought. New York: Basic Books.

Marsh, G. P. 1864. Man and Nature: or Physical Geography as modified by Human Action. New York: Charles Scribner.

Maslow, A. H. 1943. "A Theory of Human Motivation." *Psychological Review, 50, pp. 370-96.* Washington, USA: American Psychological Association.

Max-Neef, M. A. 1991. Human Scale Development. New York: Apex Press.

Mayor, F. 1995. The New Page. Paris: UNESCO Publishing.

MCEECDYA. 2008. *The Melbourne Declaration on Educational Goals for Young Australians*. Melbourne: Curriculum Corporation. [Online, accessed 15 July 2010]. URL:

http://www.mceecdya.edu.au/verve/_resources/National_Declaration_on_the_Educational_Goals_for_Young_Austra lians.pdf

MCEETYA. 1999. *The Adelaide Declaration on National Goals for Schooling in the Twenty-First Century.*Melbourne, Curriculum Corporation. [Online, accessed 8 October 2007].

URL: http://www.curriculum.edu.au/mceetya/nationalgoals/natgoals.htm

MCEETYA. 2003. *National Safe Schools Framework*. Canberra: MCEETYA. [Online, accessed 5 July 2008]. URL: http://www.mceetya.edu.au/verve/_resources/natsafeschools_file.pdf

MCEETYA. 2004. *National Assessment Program – Civics and Citizenship.* Canberra: MCEETYA. [Online, accessed 10 July 2011]. URL: http://www.curriculum.edu.au/verve/_resources/CC04_Yr_10_SRM_Part_A.pdf

McGettrick, B. J. 1995. *Occasional Paper: Values and Educating the Whole Person.* Scotland: Scottish Consultative Council on the Curriculum. [Online, accessed 23 October 2007].

URL: http://www.creativecommunities.org.uk/essays/124.html

McKeown, R. 2002. *Education for Sustainable Development Toolkit (Version 2).* USA: Waste Management Research and Education Institution. [Online, accessed 27 April 2011].

URL: http://www.esdtoolkit.org/esd_toolkit_v2.pdf

McKeown, R. 2005. *Quality Education for Sustainable Development: One route to develop and fulfil human capabilities.* USA: University of Tennessee.

Meadows, D. H. and D. L., Randers, J., and Behrens, W. 1972. *The Limits to Growth.* New York: Universe Books.

Mellor, S. 2007. "Civics and citizenship education: state of the nation." *Compak 2007, Issue 1, February, 2007.* Victoria: Victorian Commercial Teachers Association. [Online, accessed 15 July 2011]. URL: http://www.afssse.asn.au/bea/resources/Compak_feb07_5_8.pdf

Mezirow, J. 2000. *Learning as Transformation: Critical Perspectives on a Theory in Progress.* San Francisco, USA: Jossey Bass.

Monk, R. and Raphael, F. eds. 2001. *The Great Philosophers. From Socrates to Turing.* London, Phoenix: Orion Books Ltd.

Morsy, Z. ed. 1997. *Thinkers on Education (Volumes 1-4)*. Paris: UNESCO Publishing, Oxford and IBH Publishing.

Mustard, J. F. 2006. "Early Child Development Sets the Foundation for Learning, Behaviour and Health." *The Canadian Association of Principals (CAP) Volume 14, No. 3, Fall 2006.* Canada: CAP. [Online, accessed 15 April 2008]. URL: http://www.founders.net/fn/bios.nsf/cl/fn-bios-1!opendocument

Nanzhao, Z. 2005. Four 'Pillars of Learning' for the Reorientation and Reorganisation of Curriculum: Reflections and Discussions. Geneva: International Bureau of Education (IBE). [Online, accessed 24 January 2008].

URL: http://www.ibe.UNESCO.org/cops/Competencies/PillarsLearningZhou.pdf

National Geographic News. 2007. "Global Warming 'Tipping Points' Reached, Scientist Says." *National Geographic News, December 14, 2007.* California. [Online, accessed 20 July 2011]. URL: http://news.nationalgeographic.com/news/2007/12/071214-tipping-points.html

National Curriculum Board. NCB. 2008. "National Curriculum Journey Begins." *Media release, 2 May 2008*. Canberra: National Curriculum Board (NCB).

Nicolescu, B. 1996. *The Transdisciplinary Evolution of Learning.* France: International Centre for Transdisciplinary Studies and Research (CIRET).

[Online, accessed 22 April 2008]. URL: http://www.learndev.org/dl/nicolescu f.pdf

Nisbet, J. 1999. "Policy-oriented Research". Issues in Educational Research. Oxford: Elsevier Science.

Nisbett, R. E. 2009. *Intelligence and how to get it.* New York: Norton and Company.

Noddings, N. 1984. *Caring: A feminine approach to ethics and moral education*. Berkeley, CA: University of California Press.

OHCHR. 1969. *International Convention on the Elimination of All Forms of Racial Discrimination.* Geneva: UNHCHR. [Online, accessed 10 October 2007]. URL: http://www2.ohchr.org/english/law/cerd.htm

OHCHR. 1976a. *International Covenant on Civil and Political Rights*. Geneva: UNHCHR. [Online, accessed 10 October 2007]. URL: http://www2.ohchr.org/english/law/ccpr.htm

OHCHR. 1976b. *International Covenant on Economic, Social and Cultural Rights*. Geneva: UNHCHR. [Online, accessed 10 October 2007]. URL: http://www2.ohchr.org/english/law/cescr.htm

OHCHR. 1990. *Convention on the Rights of the Child.* Geneva: UN High Commission for Human Rights (UNHCHR). [Online, accessed 8 October 2007]. URL: http://www2.ohchr.org/english/law/crc.htm

OHCHR. 2006. *Plan of Action. World Programme for Human Rights Education (WPHRE), 2005-2007.* Geneva: UN High Commission for Human Rights (UNHCHR). [Online, accessed 1 May 2010]. URL: http://www.ohchr.org/Documents/Publications/PActionEducationen.pdf

O'Sullivan, E. 2003. "Bringing a perspective of transformative learning to globalised consumption." *International Journal of Consumer Studies, 27 (4), pp. 326–330.* Ontario, Canada: Ontario Institute for Studies in Education, University of Toronto.

Oxford. 2005. *Oxford Advanced Learner's Dictionary of Current English.* 2005. 7th Edition, UK: Oxford University Press.

P21. 2002. Framework for 21st Century Learning. Washington DC, USA: Partnerships 21. [Online, accessed 18 May 2012]. URL: http://www.p21.org/overview/skills-framework

Pascoe, S. 2005. "Assessing Social Competence: Can and should we be assessing in the social domain?" *Curriculum and Assessment: Closing the Gap conference, June 2005*, Brisbane: Curriculum Corporation. [Online, accessed 12 July 2011]. URL: http://web.cecv.catholic.edu.au/reseminar/conferences/assessingsc.pdf

Patsula, P. J. 1999. *Applying Learning Theories to Online Instructional Design*. Seoul: Sookmyung Women's University. [Online, accessed 28 March 2008]. URL:

http://www.patsula.com/usefo/webbasedlearning/tutorial1/learning_theories

Pethica, J. et al. 2010. Climate Change: A Summary of the Science. London: The Royal Society.

Pigozzi, M. J. 2004. "The Ministerial Viewpoint on Quality Education." *Prospects No. 30, Quarterly Review of Comparative Education Vol. xxxiv, No. 2, June 2004.* Geneva: International Bureau of Education.

Pinker, S. 2003. The Blank Slate. The Modern Denial of Human Nature. London: Penguin.

Pronk, J. and Haq, M. 1992. Sustainable Development: From Concept to Action. The Hague Report. New York: United Nations Development Programme.

Pugh, G. E. 1977. The Biological Origin of Human Values. New York: Basic Books.

Quine, W. V. and Ullian, J. S. 1970. The Web of Belief. New York: Random House.

Raths, L., Harmin, M. and Simon, S. 1966. *Values and Teaching: Working with Values in the Classroom*. Ohio: Merrill Publishing.

Rawls, J. P. 1971. A Theory of Justice. Cambridge MA: Harvard University Press.

Reid, A. 2005. *Rethinking National Curriculum Collaboration. Towards an Australian Curriculum.* Canberra: Commonwealth of Australia.

Resnick, L. B. 2010. "Nested Learning Systems for the Thinking Curriculum." 2009 Wallace Foundation

Distinguished Lecture. Educational Researcher, Vol. 39, No. 3, pp. 183–197 April 2010. [Online, accessed 10]

June 2010]. URL:

http://www.aera.net/uploadedFiles/Publications/Journals/Educational_Researcher/3903/183-197_04EDR10.pdf

Reynolds, R. 2001. *HSIE 7-10 Literature Review*. Sydney: New South Wales Board of Studies. [Online, accessed 25 July 2010]. URL: http://www.boardofstudies.nsw.edu.au/manuals/hsie_710_lit_review.pdf

Ridley, M. 2004. *Nature via Nurture. Genes, Experience and what makes us Human*. London: UK, HarperCollins Publishers.

Rokeach, M. 1973. The Nature of Human Values. New York: John Wiley.

Rowe, K. J. 2003. "The importance of Teacher Quality as a key determinate of students' experiences and outcomes of schooling." *ACER Research Conference Proceedings, Melbourne 2003, Building Teacher Quality.*Melbourne: ACER.

SACSA. 2001. South Australian Curriculum Standards and Accountability Framework. South Australia: Department of Education and Children's Services (DECS). [Online, accessed 22 April 2008]. URL: http://sacsa.sa.edu.au/index_fsrc.asp?t=ECCP

Sato, M. 2006. Evolving Environmental Education and its relation to EPD and ESD. Tokyo: Asia Pacific Cultural Centre for UNESCO (ACCU).

Saul, J. R. 1993. Voltaire's Bastards. The Dictatorship of Reason in the West. Canada: Penguin.

Saul, J. R. 1997. The Unconscious Civilization. Australia: Penguin books.

Saul, J. R. 2006. The Collapse of Globalism. Victoria, Australia: Penguin Group.

Scheerens, J., Vermeulen, C. and Pelgrum, W. J. 1989. "Generalizability of instructional and school effectiveness indicators across nations." *International Journal of Educational Research, 13 (7): 789-799*.

Schwartz, S. H. 1994. "Are there Universal Aspects in the Structure and Contents of Human Values?" *Journal of Social Issues Vol. 50 (4), pp 19–45.* Boston: Blackwell Publishers.

Scriven, M. 1966. "Student values as educational objectives." *Proceedings of the 1965 Invitational Conference on Testing Problems, 1966, pp. 33-49.* Princeton: Educational Testing Service.

Sen, A. 1999a. *Development as Freedom*. New York: Random House.

Sen, A. 1999b. "Democracy as a Universal Value." *Journal of Democracy 10.3 (1999) pp. 3-17.* USA: Johns Hopkins University Press. [Online, accessed 14 March 2008]. URL:

http://www.adelinotorres.com/desenvolvimento/Amartya%20Kumar%20Sen%20-

%20Democracy%20as%20a%20Universal%20Value%20-%20Journal%20of%20Democracy%20103.htm

Sen, A. 2000. East and West: The reach of reason. New York: Review of Books.

Shaeffer, S. 2006. "Opening address. UN Decade for Education for Sustainable Development" 6th Asia and Pacific Experts' Consultation on Reorienting Teacher Education to Address Sustainability, August 22-25, 2006. Penang, Malaysia: UNESCO. [Online, accessed 27 April 2008].

URL:http://www.UNESCObkk.org/fileadmin/user_upload/esd/documents/workshops/penang06/Sheldon_S haeffer ppt 200kb.pdf

Shaeffer, S. 2007a. "Filling a Half-Empty Glass: Learning to Live Together through Education for Sustainable Development." *Paper presented at the XIII World Congress of Comparative Education Societies, August 2007. Living Together: Education and Intercultural Dialogue.* Sarajevo: WCCE, University of Sarajevo. [Online, accessed 27 April 2008]. URL: http://www.wcces2007.ba/eng/WCCS2007.pdf

Shaeffer, S. 2007b. "Learning to Transform: Fostering Change through a Chaordic Perspective." *Keynote Address, 3rd International Conference on Gross National Happiness, World Views Make a Difference: Towards Global Transformation, 22–24 November 2007.* Nongkhai, Thailand: UNESCO.

Shannon and McCall. 2000. *Positive Learning Environments in Schools. A Pan-Canadian Consensus Statement*. Canada: Shannon and McCall. [Online, accessed 25 March 2008].

URL: http://www.safehealthyschools.org/positivelearning.htm

Slaughter, R. A. 1999. Futures for the Third Millennium. Enabling the Forward View. New South Wales: Prospect Media.

Smith, E. R, and Tyler, R. W. 1942. *Appraising and recording student progress*. Adventure in American Education Series, Vol. 3. New York: Harper.

Snow, C. P. 1963. The Two Cultures: And a Second Look. New York: Mentor.

Somerville, M. A. and Rapport, D. J. 2000. *Transdisciplinarity: Recreating Integrated Knowledge*. Oxford: EOLSS Publishers.

Sperry, R. 1983. *Science and Moral Priority. Merging Mind, Brain and Human Values.* Oxford: Basil Blackwell.

Sterling, S. 2001. Sustainable Education. Re-visioning Learning and Change. Devon, UK: Green Books.

Sterling, S. 2003. Whole Systems Thinking as a Basis for Paradigm Change in Education: Explorations in the Context of Sustainability. Bath, UK: University of Bath.

The Australian. Ferrari, J. 2010. "Curriculum 'is unclear and not ready to teach." *The Australian, 23 October* 2010.

Trivers, R. L. 1971. "The evolution of reciprocal altruism". *The Quarterly Review of Biology Vol. 46, No. 1, (March 1971): 35–57.* Chicago: The University of Chicago Press.

Turiel, E. 1983. *The Development of Social Knowledge: Morality and Convention.* Cambridge: Cambridge University Press.

UN. 1945. *The United Nations Charter*. New York: United Nations (UN). [Online, accessed 14 March 2008]. URL: http://www.un.org/aboutun/charter/

UN. 1948. Universal Declaration of Human Rights. New York: United Nations (UN).

UN. 1992. "Agenda 21: Programme of Action for Sustainable Development." *Earth Summit. United Nations Conference on Environment and Development (UNCED), 3-14 June 1992, Rio de Janeiro, Brazil.* New York: United Nations.

UN. 1993. "Vienna Declaration and Programme of Action." *World Conference on Human Rights.* A/CONF.157/23, 12 July 1993. Geneva: UNHCHR. [Online, accessed 15 March 2008]. URL: http://www.unhchr.ch/huridocda/huridoca.nsf/(Symbol)/A.CONF.157.23.En

UN. 1996. Observance of the International Year for the Eradication of Poverty and proclamation of the first United Nations Decade for the Eradication of Poverty (1997 – 2006). New York: UN General Assembly. [Online, accessed 25 April 2007].

URL: http://daccess-ddsny.un.org/doc/UNDOC/GEN/N96/762/67/PDF/N9676267.pdf?OpenElement

UN. 1999. *Declaration and Programme of Action on a Culture of Peace*. New York: United Nations General Assembly. [Online, accessed 18 September 2005]. URL: http://cpnn-world.org/resolutions/resA-53-243B.html

UN. 2000. *UN Millennium Development Goals*. New York: United Nations. [Online, accessed 26 September 2007]. URL: http://www.un.org/millenniumgoals/

UN. 2002a. Report of the World Summit on Sustainable Development (WSSD). Johannesburg, South Africa. New York: United Nations.

UN. 2002b. Resolution adopted by the General Assembly 59/237. UN Decade of Education for Sustainable Development. Fifty-ninth session, UN General Assembly. Agenda item 85 (g). New York: UN. [Online, accessed 16 March 2008].

URL: http://daccessdds.un.org/doc/UNDOC/GEN/N04/490/48/PDF/N0449048.pdf?OpenElement

UN. 2003. International Human Rights System. Human Rights in Times of Emergency, Part II. New York:

United Nations. [Online, accessed 26 January 2008]. URL:

http://www.un.org/esa/socdev/enable/comp210.htm#10.2

UN. 2005a. *Crossing the Divide: Dialogue among Civilisations. Report to the UN of the Group of Eminent Persons.* New York: United Nations.

UN. 2005b. Sixtieth Anniversary of the United Nations. History of the United Nations. New York: United Nations. [Online, accessed 23 January 2008]. URL: http://www.un.org/aboutun/sanfrancisco/members.html

UN. 2006. "Address upon receiving Alliance of Civilisations Report, Stress Basic Values Common to all Religions." *Media Release. Secretary-General in Istanbul.* New York: UN Department of Public Information. [Online, accessed 28 January 2008]. URL: http://www.un.org/News/Press/docs/2006/sgsm10733.doc.htm

UNCHE, 1973. Report of the United Nations Conference on the Human Environment. New York: United Nations.

UNDP. 1990. Concept and Measurement of Human Development. New York: Oxford University Press.

UNDP. 1999. *Human Development Report 1999. Globalization with a Human Face.* New York: Oxford University Press.

UNDP. 2007. Annual Report 2007. Making Globalization Work for All. New York: UNDP.

UNDP. 2008. *Human Development Report 2007-2008. Fighting Climate Change: Human Solidarity in a Divided World.* New York: Palgrave Macmillan.

UNESCO. 1945. Constitution of the United Nations Educational, Scientific and Cultural Organization. Paris: UNESCO. [Online, accessed 14 March 2008]. URL: http://portal.UNESCO.org/en/ev.php-URL_ID=15244&URL_DO=DO_TOPIC&URL_SECTION=201.html

UNESCO. 1995. Integrated Framework of Action on Education for Peace, Human Rights and Democracy.

Paris: UNESCO. [Online, accessed 15 April 2006].

URL: http://www.UNESCO.org/education/nfsUNESCO/pdf/REV_74_E.PDF

UNESCO. 1995a. *Declaration of Principles on Tolerance*. Paris: UNESCO. [Online, accessed 15 January 2008]. URL: http://www.UNESCO.org/cpp/uk/declarations/tolerance.pdf

UNESCO. 1997. *Declaration on the Responsibilities of the Present Generations Towards Future Generations.*Paris: UNESCO. [Online, accessed 29 March 2007].

URL: http://www.UNESCO.org/cpp/uk/declarations/generations.pdf

UNESCO. 1999a. *Universal Declaration on the Human Genome and Human Rights*. Paris: UNESCO. [Online, accessed 29 March 2007]. URL: http://www.UNESCO.org/new/en/social-and-human-sciences/themes/bioethics/human-genome-and-human-rights/

UNESCO. 1999b. UNESCO at the Dawn of the 21st Century. Paris: UNESCO.

UNESCO. 2000. *Manifesto 2000 for a Culture of Peace and Non-violence*. Paris: UNESCO. [Online, accessed 19 July 2007]. URL: http://www.UNESCO.org/bpi/eng/UNESCOpress/99-38e.htm

UNESCO. 2001. *Universal Declaration on Cultural Diversity*. Paris: UNESCO. [Online, accessed 24 October 2007]. URL: http://unesdoc.UNESCO.org/images/0012/001271/127160m.pdf

UNESCO. 2003. *A Transdisciplinary Approach to Education: An Instrument for Action.* UNESCO Information Brief on the UN Decade for Education for Sustainable Development. Paris: UNESCO.

UNESCO. 2003a. 32 C/5 Draft Programme and Budget. Paris: UNESCO.

UNESCO. 2005a. "Guidelines and Recommendations for Re-orienting Teacher Education to Address Sustainability." *Education for Sustainable Development in Action, Technical Paper No. 2.* Paris, France: UNESCO Education Sector. [Online, accessed May 10 2011]. URL: http://unesdoc.UNESCO.org/images/0014/001433/143370E.pdf

UNESCO. 2005b. Links between the Global Initiatives in Education. Paris: UNESCO.

UNESCO. 2005c. *Contributing to a more Sustainable Future. Quality Education, Life Skills and ESD.* Paris: UNESCO. [Online, accessed: 28 April 2011]. URL: http://unesdoc.UNESCO.org/images/0014/001410/141019e.pdf

UNESCO. 2006. *Universal Declaration on Bioethics and Human Rights*. Paris: UNESCO. [Online, accessed 3 July 2010]. URL: http://www.eubios.info/udbhr.pdf

UNESCO. 2010. UNESCO Strategy for the Second Half of the United Nations Decade of Education for Sustainable Development. Paris: UNESCO.

UNESCO-APCEIU. 2007. *Two Concepts, One Goal: Education for International Understanding and Education for Sustainable Development*. Bangkok: UNESCO and APCEIU. [Online, accessed 28 September 2007].

URL: http://www.UNESCObkk.org/fileadmin/user_upload/apeid/Publications/InfoSheets.pdf

UNESCO Bangkok. 2005a. *Asia Pacific Regional Strategy for Education for Sustainable Development.*Bangkok, Thailand: UNESCO Asia and Pacific Regional Bureau for Education.

UNESCO Bangkok. 2005b. *A Situational Analysis of Education for Sustainable Development in the Asia Pacific Region.* Bangkok, Thailand: UNESCO Asia and Pacific Regional Bureau for Education. [Online, accessed August 2010]. URL:

http://www.UNESCObkk.org/fileadmin/user_upload/esd/documents/esd_publications/situational_analysis .pdf

UNESCO Bangkok. 2008. *ESD on the Move: National and Sub-regional ESD Initiatives in the Asia-Pacific Region.* Bangkok: UNESCO Asia and Pacific Regional Bureau for Education.

UNESCO Dakar. 2000. *The Dakar Framework for Action. Education For All: Meeting Our Collective Commitments*. Paris: UNESCO. [Online, accessed 24 March 2008].

URL: http://unesdoc.UNESCO.org/images/0012/001211/121147e.pdf

UNESCO EB. 2000. *UNESCO in a Globalizing World*. (Item 8.5, 159th session of the UNESCO Executive Board Meeting, 159 EX40). Paris: UNESCO.

UNESCO EFA. 2005. EFA Global Monitoring Report. Understanding Education Quality. Paris: UNESCO.

UNESCO ICES. 1997. "Proceedings" International Conference on Environment and Society: Education and Public Awareness for Sustainability, 8-12 December 1997, Thessaloniki, Greece. Paris: UNESCO.

UNESCO IIS. 2004. (Draft) International Implementation Scheme for the United Nations Decade of Education for Sustainable Development (2005-2014). Paris: UNESCO.

UNESCO SD. 2005. UNESCO and Sustainable Development. Paris: UNESCO.

UNESCO-UNEP. 1978. Inter-governmental Conference on Environmental Education, 14-26 October 1977, Tbilisi, USSR. Paris: UNESCO-UNEP.

UNICEF. 2000. Defining Quality in Education. New York: UNICEF.

UNLD. 2002. *International Plan of Action for Implementing the UN Literacy Decade for Education for All.*New York: United Nations. [Online, accessed October 2007]. URL:

 $http://portal.UNESCO.org/education/en/file_download.php/f0b0f2edfeb55b03ec965501810c9b6caction+plan+English.pdf$

Vasak, K. 1977. "Human Rights: A Thirty-Year Struggle. The Sustained Effort to give Force of Law to the Universal Declaration of Human Rights." *UNESCO Courier 30:11, November 1977.* Paris: UNESCO.

VCAA. 2005. *Victorian Essential Learning Standards*. Melbourne: Victorian Curriculum and Assessment Authority. [Online, accessed 25 July 2010]. URL:

http://vels.vcaa.vic.edu.au/support/crosscurricular/values.html

Visser, J. 1999. *Overcoming the Underdevelopment of Learning: A Trans-disciplinary View.* Annual Meeting of the American Educational Research Association. Montreal, Canada: AERA. [Online, accessed 17 December 2007]. URL: http://www.UNESCO.org/education/educprog/lwf/dl/visser_f.pdf

Watson, J. D. 1968. The Double Helix: A Personal Account of the Discovery of the Structure of DNA. New York: Atheneum.

WCEFA. 1990. World Conference on Education for All. Meeting Basic Learning Needs: A vision for the 1990s. Jomtien, Thailand, 5-9 March 1990. New York: Inter-agency Commission for WCEFA. [Online, accessed 24 July 2011]. URL: http://unesdoc.UNESCO.org/images/0009/000975/097552e.pdf

Whitehead, A.N. 1929. The Aims of Education and Other Essays. New York: The Free Press.

Wilber, K. 2000. Integral Psychology. Boston: Shambhala Publications.

Willms, J. D. 2000. "Monitoring school performance for standards-based reform." *Evaluation and research in Education, 14, pp. 237-253.*

Yero, J. L. 2002. *Teaching In Mind: How Teacher Thinking Shapes Education*. Hamilton, Montana: MindFlight Publishing.

Zohar, D. and Marshall, I. 2000. Spiritual Intelligence. Great Britain: Bloomsbury Publishing.

ACRONYMS

AARE Australian Association for Research in Education

AAS Australian Academy of Science

ABS Australian Bureau of Statistics

ACARA Australian Curriculum Assessment and Reporting Authority

ACCU Asia Pacific Cultural Centre for UNESCO

ACER Australian Council for Educational Research

AEC Australian Education Council

AGPS Australian Government Publishing Service

Al Amnesty International

APCEIU Asia Pacific Centre for Education for International Understanding

APNIEVE Asia Pacific Network for International Education and Values Education

ARIES Australian Research Institute for Environment and Sustainability

ASEAN Association of Southeast Asian Nations

ASEP Australian Science Education Project

ATSI Australian Aboriginal and Torres Strait Islander peoples

AuSSI Australian Sustainable Schools Initiative

BBC British Broadcasting Commission

BCE Before the Common, Christian or Current Era

CAP Canadian Association of Principals

CCWA Curriculum Council of Western Australia

CDC Curriculum Development Centre (Australia)

CEL Center for Ecoliteracy (USA)

CIRET International Centre for Transdisciplinary Studies and Research

CPWR Council of the Parliament of the World's Religions

DECS South Australian Department of Education and Children's Services

DEET Northern Territory Department of Employment, Education and Training

DEETYA Department of Education Employment Training and Youth Affairs

DEEWR Department of Education, Employment and Workplace Relations

DEH Department of Environment and Heritage (Australian)

DESD United Nations Decade of Education for Sustainable Development (2005-2014)

DEST Department of Education Science and Training

DET NSW Department of Education and Training

DETA Department of Education Training and the Arts, Queensland

DEWHA Department of the Environment, Water, Heritage and the Arts

DIC Department of Immigration and Citizenship

DNA Deoxyribonucleic Acid (i.e. genetic code for all life forms)

DoE Department of Education, Tasmania

ECI The Earth Charter Initiative
EE Environmental Education

EFA Education For All

EfS Education for Sustainability

EIU Education for International Understanding

EPD Environment, Population and Information for Human Development Project

ESD Education for Sustainable Development

ESF Education for a Sustainable Future

GFC Global Financial Crisis

HDI Human Development Index
HDR Human Development Report

HENT Holistic Education Network of Tasmania

HIV/AIDS Human Immunodeficiency Virus and Acquired Immune Deficiency Syndrome

HPI Human Poverty Index

HRE Human Rights Education

IBE International Bureau of Education

ICE International Conference on Education

ICES International Conference on Environment and Society

ICT Information and Communication Technology

IEEP International Environmental Education Programme

IIS International Implementation Scheme for the United Nations Decade of Education for

Sustainable Development (2005-2014)

IPCC Intergovernmental Panel on Climate Change

IUCN International Union for the Conservation of Nature and Natural Resources

MCEECDYA Ministerial Council, Education Early Childhood Development and Youth Affairs

MCEETYA Ministerial Council for Education Employment Training and Youth Affairs

MDGs United Nations Millennium Development Goals

NCB National Curriculum Board

NFVE National Framework for Values Education in Australian Schools

NGO Non-Government Organisation

OECD Organisation for Economic Cooperation and Development

OHCHR Office of the UN High Commissioner for Human Rights

PDF Portable Document Format

SACSA South Australian Curriculum Standards and Accountability Framework

SD Sustainable Development

SEMP School Environment Management Plan

SOSE Studies of Society and Environment (Learning area in Australian schooling)

TELS Tasmanian Essential Learnings

UDHR Universal Declaration of Human Rights

UK United Kingdom
UN United Nations

UNCED United Nations Conference on Environment and Development

UNCHE United Nations Conference on the Human Environment

UNDP United Nations Development Programme
UNEP United Nations Environment Programme

UNESCO United Nations Educational, Scientific and Cultural Organization

UNEVOC International Centre for Technical and Vocational Education and Training

UNHCHR United Nations High Commission for Human Rights

UNLD United Nations Literacy Decade

USA United States of America

VCAA Victorian Curriculum and Assessment Authority
VEGPSP Values Education Good Practice Schools Project
WCCD World Commission on Culture and Development

WCED World Commission on Environment and Development

WCEFA World Conference on Education for All

WPHRE World Programme for Human Rights Education

WSSD World Summit on Sustainable Development

WVS World Values Survey
WWF World Wildlife Fund

LIST OF APPENDIXES

These Appendixes can be accessed at the following URL address: www.qesd.org/appendixes

Appendix 1

Values expressed in the International Implementation Scheme (IIS) for the UN Decade of Education for Sustainable Development compared to global values

Appendix 2

Scope of the four dimensions of Education for Sustainable Development (ESD) as described in the UNESCO International Implementation Scheme (IIS) for ESD

Appendix 3

Characteristics of Education for Sustainable Development (ESD) as described in the International Implementation Scheme (IIS) for the UN Decade of Education for Sustainable Development

Appendix 4

Description of Haidt and Joseph's Moral classification system

Appendix 5

Summary examples of Values in Education across time and cultures

Appendix 6

Characteristics of the UNESCO four pillars of learning

Appendix 7

The Seven Tensions of the Delors report

Appendix 8

Ten challenges for education in the twenty-first century

Appendix 9

Progression of values expressed in respective Australian National Goals for Schooling

Appendix 10

Values fostered by Australian National, State and Territory Education Authorities

Appendix 11

Guiding principles and key elements for good practice in values education

Appendix 12

Quality features in key relevant Australian values-based education documents compared to characteristics of ESD

Australian Sustainable Schools Initiative (AuSSI) Scope of Learning

Appendix 14

Benefits for students and communities of schools participating in the Australian Sustainable Schools Initiative (AuSSI)

Appendix 15

Key features of a whole school approach to Education for Sustainability (EfS) in AuSSI schools

Appendix 16

Collation of global values from 22 ESD-related international documents

Appendix 17

International documents from which global values are collated

Appendix 18

Global value sets appearing most frequently in 22 international documents related to sustainable development

Appendix 19

Application of global values to Haidt and Joseph's moral classification system

Appendix 20

Descriptive explanations of global values

Appendix 21

Global values mapped against Maslow's Hierarchy of Needs

Appendix 22

Comparative table of values in the Axial Age by era and culture

Appendix 23

The Golden Rule expressed across World Faiths

Comparison between Australian and global values and the values in the National Framework for Values Education in Australian Schools (NFVE)

Appendix 25

Comparison of quality education features from international documents relevant to values-based ESD

Appendix 26

Moral development perspectives relevant to values education for ESD

Appendix 27

Whole person competencies for Education for Sustainable Development (ESD)

Appendix 28

Issues to inform ESD curriculum content from the UNESCO International Implementation Scheme for the UN Decade for ESD

Appendix 29

Core issues for Education for Sustainable Development in the Asia Pacific Region

Appendix 30

Sustainability issues in the 2005 National Environmental Education Statement for Australian Schools

Appendix 31

Sustainability content suggested in the 2010 Sustainability Curriculum Framework

Appendix 32

Combined list of sustainability issues to inform ESD curriculum content

Appendix 33

ESD skills in the International Implementation Scheme for the UN Decade of Education for Sustainable Development

Appendix 34

Skills for Education for Sustainable Development (ESD) identified by McKeown

Combined list of sustainability skills to inform ESD curriculum development categorised according to the UNESCO four pillars of learning

Appendix 36

A paradigm shift for education in moving towards ESD in the Asia Pacific region

Appendix 37

Characteristics appropriate for ESD in the 2005 National Environmental Education Statement for Australian Schools

Appendix 38

Characteristics appropriate for ESD in Australia's 2009 National Action Plan for Education for Sustainability

Appendix 39

Transitions needed to move from traditional to contemporary quality education approaches appropriate for ESD

Appendix 40

Evaluative criteria for values-based Education for Sustainable Development

Appendix 41

Examples of references to diverse cultures in the Australian National Curriculum

Appendix 42

Examples of references to ATSI histories and cultures in the Australian National Curriculum

Appendix 43

Examples of references to Asia and Australia's engagement with Asia in the Australian National Curriculum

Appendix 44

References in the Australian National Curriculum to the environmental practices of ATSI peoples, and their connection to, and interdependence with the land

Appendix 45

References in the Australian National Curriculum to the sustainability of practices used by other cultures.

Appendix 46

Definitions of 'Reasoning' and 'Problem Solving' across year levels in the Australian National Mathematics Curriculum

Appendix 47

Frequency of skills in the Australian National Curriculum by groups of skills

Appendix 48

Full collation of skills in the Australian National Curriculum

Appendix 49

References to the 'recognition' of Indigenous peoples, and diverse cultures and languages in the Australian National Curriculum

Appendix 50

Examples of the skill of 'problem solving' in the Australian National Curriculum

Appendix 51

Examples of the use of the term 'thinking' in the Australian National Curriculum

Appendix 52

References to the use of Information and Communication Technology in the Australian National Curriculum

Appendix 53

Examples of the skill of 'decision making' in the Australian National Curriculum

Appendix 54

References to a 'futures perspective' and hypothetical phrasing in the Australian National Curriculum

Appendix 55

References to conducting experiments and making or using models in the Australian National Curriculum

Appendix 56

References to 'interconnected understandings' in the Australian National Curriculum

Appendix 57

References to 'systems' in the Australian National Curriculum

Appendix 58

References to 'transformation' and 'change' in the Australian National Curriculum

Appendix 59

Categorisation of skills in the Australian National Curriculum according to the UNESCO four pillars of learning.

Appendix 60

References to global issues in the Australian National Curriculum

Appendix 61

References to environmental issues in the Australian National Curriculum

Appendix 62

Examples from the Science learning area in the Australian National Curriculum emphasising the importance of Science for environmental sustainability

Appendix 63

Comparison between ESD issues to inform curriculum content listed in Appendix 32 and ESD content in the Australian National Curriculum

Appendix 64

References to socio-economic and socio-political issues in the Australian National Curriculum

Appendix 65

Values appearing most frequently in the Australian National Curriculum compared to those in the Melbourne Declaration, the NFVE and global values

Full collation of values in the Australian National Curriculum

Appendix 67

References to 'difference' in the Australian National Curriculum

Appendix 68

Tally of global values in the Australian National Curriculum

Appendix 69

Instances of 'interdependence' and 'interconnectedness' in the Australian National Curriculum

Appendix 70

Summary of references to the terms 'values', 'morals', 'ethics' (or ethical), 'attitudes', 'principles' and 'beliefs' in the Australian National Curriculum

Appendix 71

Full collation of instances of the terms 'values', 'morals', 'ethics' (or ethical), 'attitudes', 'beliefs' and 'principles' in the Australian National Curriculum

Appendix 72

Frequency of references relating to each ESD dimension in the Australian National Curriculum

Appendix 73

References to cross-disciplinary cooperation in the Science curriculum

Appendix 74

Opportunities to develop emotional intelligence in the Australian National Curriculum

Appendix 75

References to aesthetic appreciation in the English learning area in the Australian National Curriculum

Appendix 76

Opportunities for civic contribution or taking action towards sustainability in the Australian National Curriculum

References to religion or spirituality in the Australian National Curriculum

Appendix 78

References to relationships in the Australian National Curriculum

Appendix 79

Summary evaluation of the Australian National Curriculum against the evaluative criteria for values-based ESD in Appendix 40

ABOUT UNESCO APNIEVE Australia

UNESCO APNIEVE Australia is the Australian Chapter of the Asia Pacific Network for International Education and Values Education. APNIEVE has as its principal objectives the promotion and development of international education and values-based education for peace, human rights, democracy and sustainable development.

The Regional network was established in 1995 under the auspices of UNESCO while the Australian Chapter was established in Adelaide in May 2000. Since that time APNIEVE has been active in conducting professional development for educators and in developing a series of source books for teachers.

More information about UNESCO-APNIEVE Australia may be found at the national website:

www.unesco-apnieve.edu.au

ABOUT THE AUTHOR

Dr Joy de Leo has worked in the field of education for over 35 years in both the public and private sectors, across a variety of diverse roles, including policy and curriculum development, educational administration and teaching at all levels from schooling to vocational and higher education. Her educational experiences include working with Indigenous and culturally diverse migrant communities, international students and with students, teachers and teacher educators throughout the Asia Pacific region in Education for Sustainable Development.

Some of the many roles she has held in South Australia include Principal Policy Adviser to the CEO of Education, Regional Director of AusAID, Director of Multicultural SA, Executive Officer ACPET SA, as well as Member of the

Australian National Commission for UNESCO, UNESCO's Asia Pacific Regional Group of Experts on Education for Sustainable Development and the Australian National Green Skills Advisory Implementation Group.

Dr de Leo is the Founding President of UNESCO-APNIEVE Australia and Vice-President of UNESCO-APNIEVE in the Asia Pacific region. She has been awarded an Order of Australia Medal for her contribution to Public Service. Dr de Leo is involved in speaking, writing and conducting workshops for teachers, curriculum writers, educational administrators and teacher educators on Education for Sustainable Development in Australia and in the Asia Pacific region. She may be contacted at: apnieve@adam.com.au

Other books produced by UNESCO APNIEVE

APNIEVE. 1998. Learning to Live Together in Peace and Harmony. A Sourcebook for Teachers. Bangkok: UNESCO Publishing.

APNIEVE. 2002. Learning to Be. A Holistic and Integrated Approach to Values Education for Human Development. Bangkok: UNESCO Publishing.

APNIEVE. 2005. *Learning to Do. Values for Learning and Working Together in a Globalised World.*Bonn, Germany: UNEVOC.

APNIEVE. 2009. Learning to Know for a Peaceful and Sustainable Future. Korea: UNESCO APCEIU.

Paperback copies of this book may be purchased from UNESCO APNIEVE Australia by ordering online at: www.qesd.org